UNIVERSITY OF PÉCS FACULTY OF BUSINESS AND ECONOMICS

DOCTORAL SCHOOL OF BUSINESS ADMINISTRATION

Khaled Omar Abbas

Audit Performance and Practicing Environment Reform: Audit Practicing Environment in Egypt

DOCTORAL DISSERTATION

Supervisor: Prof. Dr. Iván Belyácz

Pécs, 2015

Abstract

This thesis presents a study to identify and measure the interrelationship between proposed audit environment's main factors, in order to present a suggested model for audit fundamentals' reform in Egypt. This model will increase the professional auditors' and practicing regulatory bodies' roles. Due to the globalization of capital markets, the audit environment becomes more complex and includes many interactive relationships between its main fundamental factors. Recent accountability failures in the United States and other countries raised the fact that the accounting and auditing professional practicing environment needs to be reformed, modernized, and expanded.

Despite the increased global attention to the importance of audit practice reforms, there are a limited number of academic researches that discussed the practicing environment changes and their affects on auditors' responsibility and operating process.

An extensive review of the literature revealed a strong relationship between audit practicing and recent financial failures, as well as the subsequent global expansion of practicing reforms.

This thesis makes an important contribution to the audit-practicing global harmonization through developing a theoretical Conceptual Practicing Environment model that has been created and follows a deductive approach, and primary data questionnaires collected from the designated practitioners in different local big audit firms. Data was analyzed according to their descriptive properties and underlying correlation structure. Several principal components were derived from these analyses, which were used in hypothesis testing, in order to examine the interrelationships between the factors associated with audit-practicing environment.

Furthermore, implications for academics and practitioners are drawn that indicate the relevance and applicability of this research to audit practice reforms. Research limitations for possible future research are set out. The thesis is organized into eight chapters, which are entitled in the following order: chapter one introduces the overview and the background of the study, chapter two introduces a review of related literatures, chapter three presents an analysis of the relationship between audit practice and recent financial market failures, chapter four examines different internal and external audit-practicing environment variables, chapter five discusses the current environment of audit practice in Egypt, chapter six proposes the suggested factors that affect the efficiency of the audit-practicing environment, chapter seven analyzes the correlation and multiple regressions based on the questionnaire and empirical tests of the hypothesis, and chapter eight provides suggestions for Egypt's local practice reforms. The magnitude of each variable in the practicing environment will be differentiated from one country to another based on the local culture and influence power of environment of different variables. The thesis confirmed that the current Egyptian local practice environment is classified as a dormant, complex, and uncertain environment that needs major reforms in order to increase its practice organizations effectiveness. Also, empirical research results raised a consensus between the selected sample answers about the importance of increasing professional accountants' and auditors' technical and judgmental capability, in addition to the necessity of increasing the non-governmental organizations' subscriptions in developing and managing the practicing profession.

Table of Contents

AbstractI				
Table of Contents III				
1. Int	roduction1			
2. Pro	oblem Identification3			
3. Res	search Aims and Objectives7			
4. The	esis Hypotheses and Model13			
5. Stru	ucture of the Thesis and Order of Presentation17			
6. San	nple Construction			
7. Hyp	potheses and Analyses			
8. Em	pirical Analysis Discussion and Results27			
9. Sun	nmary of Findings and Conclusion			
10. The	esis Contribution and Motivation			
References				
Supplementary References				
List of Publications				

Introduction

1. Introduction

The emergence of international capital markets requires high cooperation between foreign and local regulators, besides capital providers. This cooperation will support funding growth, invent new ways to manage them, reduce the complexity of the current globalized business environment, and ease the capital flows between borders. In order to increase the efficiency of capital markets, businesses need to unify the processing of international financial transactions through the use of new accounting and assurance methods, such as generally accepted accounting and auditing standards, which simplify the recording of local and foreign economic deals (**ICAA**, **1998**). In particular, consider the frequent international spread of companies' financial securities, and the complicated requirements of multi-markets securities listings, which make the companies subject to both local and foreign regulations. Most financial researches present the necessity of markets efficiency, which is essential not only in wealthy countries, but also in undeveloped nations, as markets that work with efficient and transparent systems will be globally recognized.

Current changes in global businesses set a new paradigm for the audit practicing environment, which requires an integrated international harmonization between auditpracticing regulations, taxation rules, corporate laws, and accounting standards. International harmonization represents a strong challenge, as it is hard to establish globally acceptable accounting tools that enable different governments to deal with their local and international unique specifications. This new paradigm led to a more challenging environment for both local and international audit firms, which forced auditors to expand their range of business services beyond the traditional assurance and tax advices (**Epstein & Geiger, 1994**). Additionally, these changes increased the international pressures for imposing professionalism competency, and global strategic professions' mindset that support the creation of global ethical culture, knowledge, skills and social rules for both audit firms and audit-practicing institutions.

Introduction

Recently, accounting and auditing became part of the political lobbying game. Political lobbying can be forced by government, or any of the negatively affected beneficiaries, such as companies' management, executives, investors, regulatory bodies, associations, and any other parties that have the power to influence decision makers. Accordingly, it is essential to create a strong ethical culture in audit-practicing businesses by establishing practice governing rules, policies, and increasing peers to support moral practice and ethical values (**Piotroski & Srinivasan, 2008**). Therefore, professions start to focus more on developing a proper structure of global economies that supporting the International Financial Reporting Standards (IFRS), and International Auditing Standards' (IAS) role in providing high-quality financial information, which is increased due to the recent accounting and financial failures, such as Enron (**Hegarty et al., 2004**).

Definitely, the business world's quick environmental changes and globalization led to a shift in the practicing environment, and impacted the auditors' daily practice operations. The inefficient practicing environmental structure associated with governmental passion to deregulate audit practicing and credit rating services hinders the audit basic role of any financial signaling system, and also turned the practicing environment and audit firms into oligopoly profit seekers that tried their best to cover any crisis indicators and protect their interests. Accordingly audit practicing is quite simply at a truly critical point in the history of professions, which requires an immediate global reform to strengthen, modernize, and promote adherence to high-quality accounting and auditing standards that encourage knowledge exchange, in addition to developing a strong international practicing environment that follows integrated auditing best practices, and putting public interests before personal interests (**Previts, 2005**).

2. Problem Identification

Audits have recently become more important than in the past due to the globalization of capital markets. The recent accountability failures in the United States and other countries have led to bankruptcies and restatements that harmed countless shareholders, employees and retirees. This also increased awareness about accounting and auditing professions needing to be reformed, modernized, and expanded. Moreover, the audit environment has become more complex and includes many interactive relationships between its main fundamental factors. There are no specific identifications for these factors that have the main effects on the audit environment efficiency. Many accounting-related theories focus on structuring environmental systematic coherence between accounting as a social concept and its practice. The main theories are represented in Decision Usefulness Theories, Public Interest Theory, Capture Theory, Legitimacy Theory, Stakeholder Theory, and Positive Accounting Theory. **Bogle (2000)** identifies five risks that pressurized audit firms' practicing performance and ethical values. These risks represent the main reasons for the recent financial crisis. Also, they work as warning systems that limit any future audit failures. The five pressurizing risks are summarized as follows:

- 1. Outdated accounting and auditing standards that do not cope with modern economic changes.
- 2. Increasing unethical concepts, such as earnings management and accounting creative financial engineering.
- 3. Unclear new methods of accounting management's incentives, and compensations.
- 4. The aggressive unethical uses of tax shelter schemes.
- 5. Audit profession's weak independencies' measurements and increase of auditors' conflict of interests due to new practicing alternative business structures and advisories.

Typically, there are two systems to monitor public companies' performance: the first is internally applied through using accounting standards, while the second is externally managed through the markets' regulators. Market regulators have the essential legally protected tools that enable them to limit organizations' boards and management discretions, monitor companies released significant information, take any necessary actions

Problem Identification

to protect public interests, and impose of specific financial reporting standards (Gordon, **2003**). The legal system implemented in Egypt is taken from the French legal system, which is basically the Roman law system. A limited liberalization of Egyptian accounting services has been initiated since the last decade. Accounting Practice Law number 133/1951, and its amendments, represent the main law governing professional accountants and auditors. However, the law is considered out of date by most practitioners, which stalled efforts to improve accounting and auditing practices. Egyptian audit-practicing structure only permits local firms to be licensed, however, all of the big international auditing firms are presented in the Egyptian market through partnership with local professions. Currently, small- and medium-size Egyptian audit firms are not obliged to comply with stringent standards, while big firms have to follow rigid standards due to their associations with international audit firms. Also, The Egyptian Stock Exchange does not have the legitimate authority to enforce listed companies to follow International Financial Reporting Standards for their issued financial statements. The Capital Market Authority (CMA) has some limited sanctions authorities and tools. Currently, there is a scarcity of experienced auditors in Egypt because of lack of education and training, the absence of a strong independent counsel (to grant licenses, designations, and enforce training and continuous education), and the lack of competitive advantages in the auditing profession. Egyptian auditors are not required to follow any code of ethics in line with the IFAC codes. This deficiency is accompanied by a prevalence of the closely held companies and the absence of rules associated with appointing, or changing, auditors.

Egyptian audit-practicing function does not totally conform to International Auditing Professional Standards. ESAA (Egyptian Society of Accountants and Auditors) is not functioning as a self-regulating body as per IFAC recommendations. Currently, there are many practicing licensed auditors, non-members of the Egyptian Society of Accountants and Auditors, who lack necessary professional competence for providing high-quality auditing services. Also, the local university's poor educational quality, and outdated accounting and auditing curriculums that mainly focus on preliminary topics and applications without considering the breadth and depth of the studied subjects. Egyptian auditors' independency is dramatically affected by the nature of the closely held companies' structure. Most of the Egyptian companies are family business that are controlled by family members through the majority of issued capital shares, which make it difficult for the companies' Board of Directors and key management to separate between auditing and consultancy services provided by auditors

On the other hand, the GLOBE classification of the Egyptian culture, which is summarized in the following table, indicates that the Egyptian culture revolves around the masculine community that believes in groups working with an average degree of capability to change. This community is built on respecting of hierarchy, leadership, and a low degree of potential risk planning.

	Low-score	Mid-score	High-score
Gender	Egypt		
egalitarianism			
In-group			Egypt
collectivism			
Humane		Egypt	
orientation			
Uncertainty	Egypt		
avoidance			
Power distance	Egypt		

Project GLOBE Classification of Egyptians' Societal Cultures

Source: Javidan et al., 2006, *pp*. 67–90.

In this thesis, we will try to measure the interrelationship, importance, and effect of different proposed main factors on the audit environment, in order to present a suggested model for audit fundamentals reform in Egypt. This was measured through the use of the empirical research and quantitative survey. The concept design tried theoretically to identify the main environmental constructs/variables that may affect the local auditpracticing environment based on the previously discussed literatures and researches. Also, we tried to incorporate in each construct/variable to the expected sub-dimensions that may have a similar effect on the overall practicing environment. The new measures' constructed validity was assessed using perceptual data from professional auditors and practitioners. Because neither a single approach to conceptualize the audit practice environment nor to measure it has received widespread acceptance, we have been unable to build comprehensive and coherent literature about the audit practice environment and its impact on the profession's behavior. We conceive our proposed conceptual constructs to be multidimensional, and construct general measures in our empirical research to reflect this approach.

3. Research Aims and Objectives

Audit practicing significantly affects both the community and the economy; accordingly, it should be treated as a highly regulated business. It should be properly monitored by governmental regulatory bodies, local government departments, and commissions. After several attempts of accounting and auditing harmonization, many countries still follow two different tiers of accounting systems that consist of a preliminary legalization system for listed public companies, and a domestic legalization system for nonlisted companies. From a cost-effective point of view, the two-tier accounting systems may not be convenient for non-listed private companies and public auditors. Also, it creates many obstacles in the road of global accounting.

The growing number of recent financial scandals increased academics demands of applying strong creditable audit model that associated with professional auditors' traits and effective audit environment. This environment has to be based on international standards that designed to support capital market tools, and provide proper measuring of financial reporting quality. Despite implementing of international standards and rigid laws, such as SOX, financial scandals and crisis continued. Typically, the 2008 financial crisis presented the model of financial systemic failure, which encouraged the academic researchers to claim the necessity of financial market regulation. The overconfidence of the financial system, ignoring the market systematic risks by both bankers and regulators, and the complexity of financial markets led to the failure of both credit rating agencies and auditors' warning system.

In general, audit-practicing professions had a very negative role in controlling and prohibiting the recent global financial crisis, which increased the crisis damages' size and the consequences that followed. Also, the 2008's financial crisis presents the importance of auditor's social role in protecting public interests, which represents the main object that underneath audit practicing environment.

The thesis main objective is presenting an environmental model that could be used

7

Research Aims and Objectives

to reform the Egyptian audit environmental variables, and increase the professional auditors' and practicing regulatory bodies' roles through the following suggested enhancement:

- 1- Providing institutional structures to permit the auditor to maximize their self-interest in a publicly interested way, which may lead to an explicit covenant to serve the public interests;
- 2- Increasing the local professional body hegemonic role in audit practicing constituted ideologies, as well as, enhancing the coercive and consensual approaches adopted by the government;
- 3- Measuring and increasing the practicing environment fundamental factors interactions based on the type of political, social and cultural setting. This view of the perspective is pivotal to the overall auditing practicing environment efficiency that provides a useful lens to appreciate auditing and non-auditing services;
- 4- Reshaping of audit firms practicing methodologies, as well as the formation of new moral ideologies; and
- 5- Increasing the professional auditors' practicing ability to identify and diagnose problems, through continuous developing training, technical education, and supports that will enhance audit professions' reputation in both and local prospects.

The contingency organizational theory suggests that any organization is consist of number of systems that supported by subsystems and directed by specific boundaries from its surrounding environmental subsystems. The theory analysis helped in understanding the interrelationships between the subsystems, the organization environmental patterns, and other related variable. The theory argued that organization environmental structure is the major source for organization management. Accordingly organizations always trying to interpret their surrounding environment, in order to maximize their advantages, and use the out puts knowledge to reshape their operations (**Richardson, 2009**). Based on the environmental theories we tried to study the environment phenomena and its effect on audit practicing. We proposed a conceptualization of practicing environment that consists of both internal and external environment dimensions. The environmental proposed dimensions are treated as exogenous constructs (independent variables) that affect the audit practicing system; the relationships among the components of those dimensions are not

Research Aims and Objectives

specifically hypothesized. The characteristics of local environmental and weakness points are treated as endogenous constructs (dependent variables). The conceptual model framework is a causal modeling. The concept design tried theoretically to identify the main environmental constructs/variables that may affect the local audit-practicing environment based on the discussed literatures and researches.

The study is an explanatory one. However, we tried to present it in a comprehensive approach. It is important to mention that the study has some limitations. Most of the studies related to the same subject, particularly in audit regulation filed, were mainly focus on audit technical issues. There are a limited number of audit studies that tried to investigate audit environmental variables, or measure their expected effect on the general audit practicing environment. We tried to structure the bases for building a model that can be used by future researchers to construct an initial theory that measure the relationship between audit practicing environmental variables, and their effect on audit services development; accordingly we also recommend that future studies in the same field may focus on developing an integrated theory in the field of practicing environment development.

The Following proposed practicing environmental model and the description of environmental dimensional variables supposed to help in restructuring the local profession, in addition to developing of practicing licensing qualifications, auditors' competence and independence. Also they increase the efficiency of Egypt current local practicing society, and the proposed new board of accountancy.



Conceptual Practicing Environment Design: Internal and External Environment Dimensions

• Source: Researcher

The following table summarizes the definition of environmental dimensional variables

Environmental var-	Description
iables	
Legal regime and as-	Legal regime is a vital factor that supports audit creditabil-
sociated support of	ity. However, legal affects may be varied across different

non onting anolity	local cultures many literatures agree that weak low enforce
reporting quality	local cultures, many literatures agree that weak law enforce-
standards	ment will negatively affect accounting and auditing report-
	ing quality.
Regulatory bodies	Regulatory influential power is significantly influencing au-
influence over main-	dit profession. It can support the profession independent so-
taining of social and	cial and ethical trust. Also, it may help in positively devel-
ethical values	oping auditors traits
Auditors and audit	Both auditors and audit firms have a significant contribution
firms specialization	in weakening the surrounding audit practicing environment.
	In order to overcome the structured environmental deficien-
	cies we need to set up additional accountability tools, define
	auditors' duties, qualifications, techniques, as well as, stand-
	ardized procedures.
International and	Each country has its own unique practicing characteristics
local practicing	that based on many environmental factors, such as domi-
standards	nated culture, legal system, regulatory body, capital market
	structure, and other external influences. Both of local regu-
	lating systems, and international practicing standards en-
	force the profession local control matrix.
Practicing organiza-	Culture represents an interaction between human and non-
tions and profes-	human resources. Therefore, practicing organizations, such
sion's culture	as audit firms, professional associations, and governmental
	regulatory bodies must have a collective unified culture that
	fit with surrounding local, as well as, international society
	culture.

In general, both contingency and Institutional theories are
considered the most convenient environmental theories that
suitable for audit profession.
Professional practicing organizations have unique and dis-
tinct design, therefore researchers classified audit practicing
environment as complex structure that requires more indi-
vidual and grouping control vehicles. Accordingly, integrat-
ing audit practicing environment perceptual into a signal
framework will enhance its efficiency and effectiveness. In
order to structure this single framework there should be a
clear identification of audit practicing environmental varia-
bles, as well as, specific diagnose of its operating conditions.
Such of this approach will help in outlining a unitary stable
practicing environment that can be globally uniformed.

4. Thesis Hypotheses and Model

Capitalism regulations are based on competitive free-market mechanisms rather than relying on the governments (Windsor & Warming, 2009). The free market system encourages organizations to focus on competition, efficiency, and creativity rather than equality, fairness, and justice, which led to many financial wrongdoings. On the other hand, the international audit-practicing business typically represents a traditional oligopoly market that is controlled by a limited number of audit firms. Also, the financial risk assessment and valuation business is manipulated by three main international rating agencies. The inefficient financial environment structure that is based on weak laws, accounting standards, and codes of ethics encouraged market players to mask their economic performance and manipulate the facts as they wish, which led the world to the recent financial failures and economic crisis.

Often, audit-practicing ideology serves as a culture starting point for both practicing regulatory organizations and audit firm. The culture of a practicing regulatory organization represents its personality that manages values, norms, and behaviors. It can be described as a system that acts upon feedback from society, professions, and laws. This type of culture should be based on values, such as society service and public interest protection. The accounting and auditing profession environment is based on a mix of "associationism," "corporatism," "liberalism," and "legalism," which is mainly associated with the three principles of social orders: market, state, and community (**Baker et al., 2005**). The deregulation of financial services represents a key competitive environmental challenge for professions. Deregulation requires shifting of a current stable culture and environment to an adaptable ones that support the organizational society concept. On the other hand, the accounting practicing preventive environment will be more effective if it is based on authoritative compliance systems. This type of environment is always classified as a high-risk environment, which should be associated by general ethical framework that acting as a proactive tool for professions to protect auditors from any improper actions (**Heninger, 2001**). Also, it supports the reformed role of both governmental and non-governmental regulators in developing an effective audit practicing environment.

Typically, Culture instrumental practicing tools support controls and improve audit-practicing performance therefore, it is essential to understand professions cultural classification that may affect professional environmental dimensions. Academics describe culture as "collective consciousness," "collective programming," or "underlying shared assumptions" that identify a group of values (**Sackman, 1991**). Unified culture requires a consensus on a number of norms and values that have to be clearly understandable and approved by the audit-practicing community. The four main characteristics type of professional organizational culture have central tendencies that allow them to be categorized into one of the four culture models, described in following Table

Culture Models	Description	Suitability to Audit- Practicing Environment
Competence	 -Group contribution is less important than individual contributions; - More focus on the quantitative aspects than the qualitative ones; and - Achievements are driven through organizations. 	No
Control	Associated with systems, procedures, and processes that are emphasized on the followed plans.	Yes
Collaboration	 Following of team-focused culture; Focus on the in-depth knowledge of customers and stakeholders; and Create strong relationships with other organizations. 	Yes
Cultivation	Build on individuals' beliefs that deliver dramatic innovation.	No

Culture Models Classification

Source: Researcher

Also, there are four main culture types that should be considered while studying the Audit-Practicing Environment dimensions. Those culture types are classified in the following table

Culture Type	Description	Suitability to Audit-Practicing Environment
Clan culture	Applicable for flexible organization that focuses on internal maintenance, and sensitive for people and customers with high degree of commitment. The organization emphasizes human resources development, cohesion, and morale.	Yes
Hierarchi cal culture	Encourage stability, control enforcement, formal task settings, strict rules, and obedience.	Yes
Adhocracy culture	Focus on dynamic flexible external positioning, and individualism open system. The long-term objective is growth and developing new resources.	No
Marketin g culture	Focus on aggressively competing, customers' satisfaction, and product effectiveness.	No

Source: Researcher

In this research we tried to present an empirical examination of the validity and correlation between suggested five key constructs (variables) of the general audit practicing environment, and their relationship with our suggested hypotheses. The underlying assumption beyond our argument was mainly that there are some main constructs (variables) that should be seen from the point of their interrelation, and the overall general audit practicing environment. This concept can be considered as a government public management policy that improves the overall performance of both audit practicing and local economy. However, we believe that the correlation and the magnitude of each construct (variable) will be different from one country to the other based on its local fundamentals, such as community culture, structure, and economy sophistication. Accordingly, we suggest that future researches can study the comparison between these constructs within different countries. Therefore, the thesis studied the following main four hypotheses:

- 1st. Updating of practicing law will partially participate in developing audit environment;
- 2nd. Increasing the efficiency of audit-practicing environment requires structure of all environment fundamental factors / variables at the same time but not with the same degree of importance;
- 3rd. Implementing of practicing firm's quality-control procedures will enhance the efficiency of audit-practicing environment faster than other fundamental factors / variables;
- 4th. International environment fundamental factors / variables will have a positive effect on practicing environment reform; and
- 5th. The primary factors that are expected to have the most important effect in practicing environment reform will be the governmental regulatory bodies and non-governmental associations.

5. Structure of the Thesis and Order of Presentation

Chapter one includes an introduction to the research, problem identification, hypotheses, and research objectives, while Chapter Two introduces a review of related literatures to present a background introduction.

Chapter Three presents an analysis of the relationship between audit practice and recent financial market failures. The chapter considered the effect of creative accounting tools, Enron scandal, and the recent global financial crisis on the governing audit-practic-ing environment and ethical values.

Chapter Four examines different internal and external audit-practicing environment variables. It describes the effect of internal environment factors on practice organization objectives, structure, and governing culture. Also, it considers the effect of external environmental factors, such as regulators, economic, social, and practicing standards. The chapter also considers the classification of environmental uncertainty.

Chapter Five discusses the Current Environment of Audit Practice in Egypt by reviewing business laws currently in use, different governmental, non-governmental regulatory bodies, and other audit practicing organizations, while Chapter Six proposes the suggested factors that affect the efficiency of the audit-practicing environment. These factors include Non-Governmental local and international organizations, regulatory bodies, audit firms' management, and auditors' characteristics.

Chapter Seven analyses the correlation and multiple regressions based on the questionnaire and empirical tests of the hypothesis. It also discusses the statistical analysis and research results.

Chapter Eight provides suggestions for Egypt's local practice reforms through developing a new conceptual framework and practicing quality measurements. Also, it concludes the end results related to Egypt's local practicing environment.

Sample Construction

6. Sample Construction

Increasing the quality of governmental, non-governmental associations, and audit firms cannot be achieved without considering others' environmental factors that may affect their performance. Our research has been built based on studying the pre-described variables that surround both the local and international audit-practicing environment.

In order to empirically examine our hypotheses, we defined the main variables that may have effects on the audit-practicing environment performance and efficiency. A detailed survey questionnaire was designed, comprised of 51 questions, and includes five parts, each of which includes detailed questions that measure different variable implications and their relationship with the Egyptian local audit-practicing environment. These questions are measured through auditors, selected samples, and responses to agreement score of certain statements that discuss the variables effects.

Due to the importance of the study in reforming of the audit practicing environment, and audit law and regulations in Egypt, the survey was based on the highest level of professional auditors in Egypt that mainly represented in the fellows and members of The Egyptian Society of Accountants and Auditors (ESAA). Accordingly the questionnaire were mainly assumed to be distributed through interviews to all the fellows of Egyptian Society of Accountants and Auditors (ESAA); as representative of the most qualified and experienced category of audit professions.

The total sample is represented by the total population of fellows of Egyptian Society of accountants and Auditors, based on data collected from the National Institute of Accountants and Auditors the fellows total numbers was 234, based on 2008 available data. This number was taken as the target population for this study, out of which 105 fellows' auditors approved to share in the research. ESAA Fellows were contacted by telephone and letters to generate interests and obtain approval for interview. Due to the nature of accounting and auditing business filed, it talks a lot of time to fix interview appointments with the selected ESAA's fellows; however most of them asked to receive a copy of the questionnaire before approving on interview appointment. Some of ESAA contacted fellows refused to be interviewed; typically they gave reasons, such as they were too busy or they never participate in surveys.

Most of the interviews (103) have been conducted in auditors' offices while 2 interviews have been done during the international conference organized by the Syndicate of Accountants on the topic "How to Reform Auditing Environment" conducted in Egypt in February 2008 and attended by many Egyptian and Arab key professional accountants and auditors. The questionnaire is also distributed to all the conference attendants as a part of the conference tool kits.

About half of the interviews were conducted in 2 of the biggest audit firms in Egypt:

- KPMG (Hazem Hassan); and
- Price Water House.

As we previously mentioned, the number of questionnaires sent, and fellows targeted to be interviewed were 234, which represents 100% of ESAA Fellows published list during 2008. From this number only 105 fellows approve to cooperate, and interviewed, this sample size of 105 allowed for a confidence level of 95% and confidence interval of +7.72%.

Study sample demographics included 105 leading auditors' members of ESAA, 23% of them had "CPA", 10% "CIA", 3% "ACCA", 2% "CA" while others had "CIPA, CMA, CFM, and CICA". Currently, 53% of the sample were "Employees in a private auditing office", 24% "Professional Firm Owners", 4% "Governmental Auditors" and 23% work in "others; as Financial Managers or University Professors".

Regarding the work field, 75% of the sample work in "auditing", 34% "general accounting", 29% "taxing", 6% "costing", 5% "insurance" and 8% "others; as training, consultancy, fidelity studies, financial management". 39% of the interviewed auditors

Sample Construction

evaluated their professional performance as "excellent", 55% "good", 5% "average" and 1% "bad".

7. Hypotheses and Analyses

	Hypothesis De-	Empirical Results Summary
	• 1	Linpirical Results Summary
First Hy- pothesis	Hypothesis De- scription Updating of practicing law will partially par- ticipate in devel- oping audit envi- ronment.	Empirical Results Summary Empirical results support the first hypothesis due to the followings : -Based on the agreement score analysis updating practicing law ranked as the second in importance comparing to the importance of developing the mutual cooperation between the governmental au- thorities and professional associations. -However, practicing law is considered as one of most important factors in the practicing environ- ment, it also came in the second level, as "The laws regulating the auditing process and the audi- tor's role" has a high ranking score of 1.39. It was less in environmental importance than "Interna- tional auditing standards" that has the highest en- vironmental ranking score of 1.33. - Weak (0.301) positive coefficient correlation be-
		–
		 dit practicing, the coefficient correlation results were (-0.653). Non-governmental local professional organizations, regulating audit practicing, have a positive coefficient correlation with outdated Practicing Law (0.492). KMO and principle component analysis of gen
		- KMO and principle component analysis of gen- eral environment variables shows a high score of profession outdated law and rules with score (.720), and the associated lack of auditing stand- ards (.604). While the laws regulating the auditing has a negative value of721, which indicate that it is uncorrelated to the general practicing envi- ronment. Accordingly it supports our hypothesis

		that law partially affect on practicing environment reform.
Second Hypothesis	Increasing of the efficiency of au- dit practicing en- vironment re- quires structure of all environ- ment fundamen- tal factors in the same time but not with the same degree of im- portance.	 Empirical results support the second hypothesis due to the followings : The importance of each environmental factors were indexed as follow : International auditing standards, (ranking score of 1.33). The laws regulating the auditing process and the auditor's role, (ranking score of 1.39). The efficiency of the professional audit firms procedures, (ranking score of 1.61). International financial reports standards, (ranking score of 1.66). The supervisory governmental organizations controlling auditing quality, (ranking score of 1.75). Non-governmental professional organizations that regulating the audit practicing, (ranking score of 1.77). Non-governmental international organizations that regulating the audit general environment, (ranking score of 2.48). KMO and principle component analysis of general environment variables shows an importance ranking of environment components as follows:
		 non-governmental international organiza- tions regulating the auditing profession (.844)
		• non- governmental professional organiza- tions regulating the auditing work (.805)
		 international financial reporting standards (.804)
		• The supervisory governmental organiza- tions controlling auditing works (.731).
		• International Auditing Standards (.581)

		 Efficiency of professional organization (.510) The laws regulating the auditing process and the auditor's law (721)
Third Hypothesis	Increasing of au- dit practicing firms' quality control proce- dures will in- crease the effi- ciency of audit practicing envi- ronment rapidly than other funda- mental factors.	 Empirical results do not support the third hypothesis due to the followings : The efficiency and quality of the professional audit firms' procedures has ranked as the third variable that affect audit practicing environment, (ranking score of 1.61). Statistical analyses confirm the necessity of developing auditors' competency. Therefore, implementation of both International Auditing Standards and updated of practicing law may have a higher ranking weight, and stronger magnitude in improving the audit practicing environment. The Pearson correlation result was positive (.158). The Pearson correlation of developing the clients' audit firms services' procedure presents a positive relation that could enhance practicing environment (.138) The Pearson correlation of staff periodical update for any new practicing environment (.209) The Pearson correlation analysis was positive for "the professionalism of the auditor; as one of the factors that affect audit practicing environment" (0.232) There is a weak positive coefficient correlation of (0.492) between the efficiency of audit firms' followed procedures, and non-governmental professional organizations that regulate audit practicing; KMO and PCA of general environment variables show a high score of both inconvenience of Egyptian auditing environment (ranking .706), and the

		auditors' limited responsibility towards the bene- ficiary (ranking .740).
Fourth Hy- pothesis	International en- vironment funda- mental factors will have a posi- tive effect in practicing envi- ronment reform.	Empirical results partially support the fourth hypothesis due to the followings : -"International auditing standards" with ranking score of (1.33), has effect on the auditing environment more than "International financial reports standards", (ranking score of 1.66). While Nongovernmental international organizations that regulating the audit practicing general environment have the least effect on practicing environment, (ranking score of 2.48)Both International Auditing Standards and International Financial Reporting Standards have a strong positive coefficient correlation with the lack of local auditing standards, and the outdated Practicing Law. The results were (0.465, and 0.419), respectively There was a strong positive coefficient correlation between International Auditing Standards, and non-governmental international organizations, that regulate audit practicing, with a correlation factor of (0.578, and 0592) - The coefficient correlation of (.0565)between non-governmental professional organizations that regulate local practicing, and non-governmental international organizations for (.0565)between non-governmental professional organizations that regulate local practicing and non-governmental international organizations for (.844 & .804) respectively. The importance in dex may be different based on
pothesis	tors that expected to have the most	the country level of development and culture. For example developed country, such as US, UK and

important effect in practicing en- vironment re- form will be the governmental regulatory bodies and non-govern- mental associa- tions.	Canada, have non-governmental practicing regu- lators that manipulating the practicing environ- ment more than any other environmental factor. This was supported by the KMO and PCA of gen- eral environment variables that show a high score for non-governmental international organizations regulating the auditing profession (score .844), and both of non- governmental professional or- ganizations regulating the auditing work (score .805) and the supervisory governmental organiza- tions controlling auditing works (score .731).
	Even though, Empirical results of Egyptian local practicing environment do not support the fifth hypothesis due to the followings :
	 Both governmental regulatory bodies and non-governmental associations were locally ranked less important in their effect on local audit practicing environment than other variables, (ranking score of 1.75, and 1.77) respectively. Correlation analysis presented a weak relationship between the efficiency of audit firms' internal practicing procedures and both of local non-governmental and governmental organizations, those responsible of regulating the audit practicing. The correlation factors were (0.312, and 0.042) respectively Correlation between staff of audit practicing firms, and both of local non-governmental, and governmental organizations that regulate practicing was weak, (0.337, and 0.060) respectively The correlations between audit clients and both of local non-governmental organizations that regulations, which regulating practicing, were positively weak, (0.435, and 0.294) respectively Non-governmental international organizations that organize audit practicing have a positive re-
	gression factor of (0.469), which is higher than the positive (0.387) regression factor of local non- governmental professional organizations. This is due to weakness and the ineffective role of local non-governmental organizations.

	- The government supervision and controlling or- ganizations, which regulate audit practicing, have a positive regression of (0.539). However, their coefficient correlation indicates a negative rela- tion with both outdated Practicing Law and lack of auditing standards. Also, it indicates a positive relation with the easiness of obtaining practicing license. The coefficient correlation results were (- 0.653, -0.447, and 0.553), respectively.
--	---

8. Empirical Analysis Discussion and Results

Researchers in accounting literatures have tried to study individually each of our proposed practicing environment constructs (variables). Many researchers studied the process of auditors' ethical decision while others studied the effect of laws and regulations on audit practicing, but none of them have measured the correlations, and the effect of those overall variables on the overall audit environment.

Based on our proposed conceptualization of practicing environment that consists of both internal and external environment dimensions, the following table presents a summary of internal environment correlation and regression analysis:

Pearson correlation	Correlation	
Q 10.1	q9_6	Q10.1
		1- The
		efficiency
	f-	of the
	Inconvenien	profession
	cy of	al
	auditing	organizati
	environmen	•
	t	procedures
The effect of the professional organisations practicing auditing on the auditing environment		1
1- The mutual understanding between the customers and the professional organisations on the		
provided services helps improving the performance of audit practicing environment.	0.138	0.165
2- The availability of a system for evaluating the efficiency and quality of the relation with the		0.100
customer has a limited effect on the auditing job environment.	0.115	0.02
3- The planning for the control arrangements of applying the professional quality in a way that		0.02
meets the organisational requirements and technical standards helps improving the performance of		0.450
		0.158
4- The auditor's awareness of the customer's work field and activities has a limited effect on the5- Keeping the auditors up to date with the changes of the professional requirements (accounting		0.04
standards, independency, objectivity, policies and procedures) helps activate the auditing job	0.066	0.209
6- The availability of internal rules at the professional organisations guarantying the regular rotation		
of auditors will help in improving the audit profession environment.	0.041	0.091
7- The organisations' use of an effective mechanism to face the failure in the application of quality		
procedures and policies helps improve the auditing job environment.	0.017	0.135
8- Do you think the professional organisations practicing in the auditing field are one of the factors		
that influencing the auditing practicing environment?	0.002	0.026
The effect of the auditors' qualification on the auditing work environment		
1- Do you think that the application of continuing vocational education will be more effective if		
carried out through obligatory instructions from the professional associations regulating the auditing		0.097
2- Do you think that the professionalism of the auditor is one of the factors affecting the audit		0.066
3- Assigning the professional associations to supervise the auditor's commitment to the continuing		
education requirements will have more effect on the audit practicing environment rather than		0.166
4- Determining the auditor's annual duties should be based on the vocational training previously	0.178	0.07
5- The commitment to Code of Ethics plays a more effective role in improving the audit practicing		
environment than the laws regulating profession practice and companies work.	0.029	0.12
The effect of the Stakeholders on the audit practicing environment		
1- Do you think that the interested parties and beneficiaries (Stock companies managements, banks,		
governmental and supervisory authorities) are one of the factors affecting the audit practicing	0.073	0.03
2- Do you think that the main interest of the beneficiaries is to obtain the financial statements,		
information summary and the financial policies rather than obtaining efficient recommendations or		
realizing the scope of work and procedures applied by the professional authorities?	0.105	0.175
3- Do you think that the companies' managements growing awareness of their social responsibilities		
helps increasing the performance and efficiency of the audit practicing environment?	0.221	0.036
4- The meeting of the beneficiaries' unusual requirements is one of the auditing priorities.	0.086	0.132
5- The beneficiaries' continuing interest in following up the audit notes and recommendations		
provided by the professional audit firms and the verification of these recommendations will help		0.016
6- The management philosophy and operating way may affect the audit practicing environment.	0.163	0.019
7- Obliging the commercial organisations by the governmental supervisory authorities to apply the		0.013
code of ethics through the business ethics laws and the auditing committees' duties strongly affects		0 177
code of cames among the business cames laws and the auditing committees duties strongly affects	0.079	0.177

The study classified the main internal environmental variables that may affect the audit firms into three:

1st Practicing internal procedures;

2nd Practicing recourses (auditors' qualification); and

3rd Stakeholders / audit firms clients.

SPSS software was used in analyzing both of correlation and regression analysis for these variables that related to local audit practicing firms. Following is a summary of analysis main results:

1st Practicing internal procedures

Analysis results confirmed that local audit firms are not implementing any procedures to agree with their clients about the expected services. This weakness is affecting negatively the audit firms practicing with a Pearson correlation (.165). However, the Pearson correlation of this procedure presents a positive relation that could enhance practicing environment (.138), which propose that local audit firms are neither focusing on the types of services they provide to their clients, nor following the pre-engagement evaluation procedures that are required by international auditing standards.

Regarding audit firms implemented internal quality procedures and their effect on the practicing environment, the Pearson correlation result was positive (.158). However, the result was (.176) in relation to actual implementation of these procedures by local practicing audit firms, which lead us to conclude that most of local audit firms do not follow any internal quality procedures

Regarding following of internal procedures for developing audit staff knowledge and skills, the Pearson correlation was (.218), which confirms the none following of these procedures. The Pearson correlation result was positively weak (.040) in relation to the effect of these procedures on enhancing practicing environment. On the other edge, the periodical update of staff for any new practicing standards' changes was positively correlated to enhancing the professional organization practicing environment with Pearson correlation (.209)

2nd Practicing recourses (auditors' qualifications)

The Pearson correlation analyses shows a positive relationship between auditors following of both code of conducts, auditing standards requirements to monitor continuous education, and enhancing audit firms internal practicing environment (0.120, and 0.166), respectively. On the other hand, the Pearson correlation analysis was positive for "the professionalism of the auditor; as one of the factors that affect audit practicing environment" (0.232), which confirms the importance of developing staff professionalism, and their strong relation to enhance practicing environment.

3rd Stakeholders / audit firms clients

Pearson correlation results related to clients' internal control procedures was negative (-0.177), which indicates a weak relationship between clients implementation of code of ethics and internal control procedures, and enhancing of audit practicing environment.

In relation to increasing of clients' understanding of their social responsibilities, and enhancing of audit practicing environment, the Pearson correlation was positive (0 .222).

Also, we tried to analyze the relationship between both of audit firms internal environment variables, and audit practicing environmental variables through measuring the correlation between both of these variables, the analysis results are summarized as follows:

1st Practicing Internal Procedures

Correlation analysis presented a weak relationship between the efficiency of audit firms internal practicing procedures and both of local non-governmental, and governmental organizations, those responsible of regulating the audit practicing. The correlation factors were (0.312, and 0.042) respectively, which indicates the limited role that those organizations play for regulating, monitoring, and control local audit practicing firms. Those organizations' role is critique, and needs to be more active to comply with international auditing standards requirements, mainly, if we considered the key role of audit practicing firms and professional organization in both of economy and society.

2nd Practicing recourses (auditors' qualifications)

Correlation between staff of audit practicing firms, and both of local non-governmental, and governmental organizations that regulate practicing was weak, (0.337, and 0.060) respectively, which indicates that both of local non-governmental and governmental organizations have no effect on enhancing audit practicing professions or the practitioners.

3rd Stakeholders / audit firms' clients

The correlations between audit clients and both of local non-governmental and governmental organizations, which regulating practicing, were positively weak, (0.435, and 0.294) respectively. Also the correlation between clients, and practicing laws that regulate audit was weak (0.251). These weak correlations' results confirmed that internal local practicing environment is Stale / Dormant and need major reforms in order to increase its operating organizations effectiveness.

The following table presents a summary of external practicing environment correlation, and regression analysis:

Regression & Coefficients Analysis for local Practicing Environmental Weakne(Coefficie/Coefficie/Coefficien/Coefficie/Coefficie/Coefficient/Coefficien	Coefficie	Coefficie	Coefficien	Coefficien	Coefficie	Coefficie	Coefficient	Coefficier	legression
	q9_1	q9_2	q9_3	q9_4	d9_5 (q9_6	q9_7	010.1	
			c. The				g- The	l. The	
	-	*	Profession	d- The	ф.		uted role	efficiency	
		Defects	Practicing a	auditor's	Easmess		of the	of the	
		in some	Law and	limited	eť	Inconveni	inconveni authorities	profession	
	÷ The	of the	its	responsibili obtaining	obtaining	ency of 1	regulating	<u>a</u>	
	lack of	applied	regulating 1	ty towards 1	đie	anditing	accounting	organizatio	
	andiing	anditing	nles are i	the 	practicing (environn	and auditing 1	а	
	standards	strandards	outdated.	beneficiary	license.	ŧ	works [procedures	
a- Laws and Procedures:									
 The efficiency of the professional organization procedures 	0.081	-0.174	0.161	-0.183	-0.066	-0.361	0.084	0.000	0.342
The laws regulating the auditing process and the auditor's role	-0.061	0.016	-0.187	-0.148	0.285	0.129	0.067	-0.114	0.279
b- Organisations:									
1- Non governmental professional organisations regulating the auditing work	-0.140	0.057	-0.492	0.244	-0.306	0.174	0.306	-0.355	0.387
The supervisory governmental organisations controlling auditing works	-0.447	-0.097	-0.653	0.203	-0.553	0.163	0.541	0.206	0.539
Non governmental international organisations regulating the auditing profession	-0.556	0.114	-0.475	0.021	-0.349	0.081	0.361	0.011	0.469
c- Standards:									
 International auditing standards 	-0.086	-0.160	-0.466	-0.104	-0.160	-0.123	0.121	-0.139	0.329
2- International funancial reporting standards	0.419	-0.024	-0.140	-0.099	-0.281	-0.279	0.438	-0.127	0.353
_	_	-	_	_	-	-	-	-	-

We have empirically analyzed the regression analysis for both current weakness elements in the Egyptian practicing environment, and their effect on the audit-practicing
environment. The results proposed the major environmental variables that have a strong impact on those weakness elements. The following is a summary discussion of these variables:

- A. The government supervision and controlling organizations, which regulate audit practicing, have a positive regression of (0.539). However, their coefficient correlation indicates a negative relation with both outdated Practicing Law and lack of auditing standards. Also, it indicates a positive relation with the easiness of obtaining practicing license. The coefficient correlation results were (-0.653, -0.447, and 0.553), respectively. This confirms the necessity of increasing the role that government professional organizations play in developing the local practicing environment, which can be achieved through using the government regulatory authorities in order to reorganize the weight of local environmental variables.
- B. Non-governmental international organizations that organize audit practicing have a positive regression factor of (0.469), which is higher than the positive (0.387) regression factor of local non-governmental professional organizations. This result represents a deviation from the actual practicing model in developed countries, as local non-governmental professional organizations always play the key role in profession regulating. Accordingly, we assume that the current statistical analysis results are due to weakness and the ineffective role of local non-governmental organizations.
- C. Non-governmental local professional organizations, regulating audit practicing, have a positive coefficient correlation with both outdated Practicing Law and audit firms' practicing procedures (0.492, and 0.354). This indicates the need to have a stronger role of non-governmental local professional organizations to support practicing environment reform procedures.
- D. Both International auditing standards and International financial reporting standards have a strong positive coefficient correlation with the lack of local auditing standards, and the outdated Practicing Law. The results were (0.465, and 0.419), respectively, which confirms the important role that both auditing and accounting reporting standards can play a part in developing the practicing local environment. The results also indicate the necessity of considering those standards as an integrated part of any practicing law reforming procedure.

We also used the SPSS to prepare KMO (Kaiser-Mayer-Oklin) and PCA (Principle Component Analysis) for the two general type tested questions about the expected weighted average and importance of the practicing environmental main three factors (laws, practicing related organizations, and practicing related standards), as well as, the currently expected deficiencies in the Egyptian practicing environment. These analysis are used to support the other used statistical techniques during the data empirical study. The following table presents the Total Variance Explained

Component	Initial Eigen values		Extraction	Rotation Sums of Squared Loadings(a)			
Co	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %	Total
1	3.588	25.628	25.628	3.588	25.628	25.628	3.522
2	2.004	14.312	39.940	2.004	14.312	39.940	1.901
3	1.387	9.908	49.848	1.387	9.908	49.848	1.524
4	1.179	8.421	58.269	1.179	8.421	58.269	1.306
5	1.124	8.029	66.297	1.124	8.029	66.297	1.309
6	.938	6.701	72.998				

Total Variance Explained

However, the tested sampling size is not that big, only 105 sample units tested, the factoring analysis technique is still visible due to the tested variables linear correlation. The KMO value is more than 0.6. Also, the Bartlett test significant value is less than .05, which indicates that factor analysis is appropriate for the tested data. The tested data plot shape is systematic and taking a horizontal trend as presented in the following Scree Plot figure.

Scree Plot figure

Scree Plot



The Eigen values test, Eigen values >1, resulted in generate five main components that represents 66% of cumulative variances. In order to increase the confidence degree in the generated components we used different software, Monte Carlo PCA, in order to computes a random parallel analysis criteria for SPSS generated Eigen values. Based on this test we decided to discuss the components that have SPSS generated Eigen values more than those randomly generate by Monte Carlo PCA. Accordingly, we discussed only the first three components. The following table of Pattern Matrix presents the comparison between SPSS and Monte Carlo PCA Eigen values.

		Component					
	1	2	3	4	5		
q10_b3	.844						
q10_b1	.805						

q10_c2	.804				
q10_b2	.731				
q10_c1	.536				
q10_a1	.510	.396	372		
q9_4		.740			
q9_6		.706			
q9_3			.720		361
q9_1		503	.604		
q9_7	334	.301	.508		
q9_5	.311			.776	
q10_a2				721	
q9_2					.879

Extraction Method: Principal Component Analysis.

Rotation Method: Oblimin with Kaiser Normalization.

a Rotation converged in 19 iterations.

Also, the following table presents component correlation matrix, which shows a low degree of correlation between the different factors results, therefore the factors rotation test was not necessary.

Component	1	2	3	4	5
1	1.000	.049	.122	087	023
2	.049	1.000	005	065	.152
3	.122	005	1.000	.070	.008

Empirical Analysis Discussion and Results

4	087	065	.070	1.000	.033
5	023	.152	.008	.033	1.000

Extraction Method: Principal Component Analysis.

Rotation Method: Oblimin with Kaiser Normalization.

We analyze KMO and PCA Results based on the following structure matrix table:

Structure Matrix

	Component				
	1	2	3	4	5
q10_b3	.849				
q10_b1	.809				
q10_c2	.784				
q10_b2	.730				
q10_c1	.581				
q10_a1	.481	.468	312		.325
q9_6		.725			
q9_4		.725			
q9_3	.325		.731		336
q9_1		453	.642		
q9_7		.313	.481		.343
q9_5				.741	
q10_a2				738	

q9_2				.870	
------	--	--	--	------	--

Extraction Method: Principal Component Analysis.

Rotation Method: Oblimin with Kaiser Normalization.

The First Component

The first component includes the different types of practicing organizations that monitoring and control the audit profession. The highest loading value is associated with non-governmental international organizations that influencing the audit practicing with a weighted score of 0.844. While the second loading highest value, with weighted score of 0.805 is in favor of the local supervisory non-governmental organization that monitoring and control the audit profession. International reporting standards have the third ranking within the first factor with weighted loading score of 0.804. While the lowest loading ranking is associated with the local governmental organizations that regulating audit practicing with a score of 0.731.

The component structure reflects the importance of governing audit practicing organizations in shaping of an idle professional environment. In contrary, the Egyptian local practicing environment is ranking these governing organizations in a lower environmental value due to the inconvenience of the local environment. Both of the Egyptian governmental regulatory bodies and non-governmental organizations are ranked as lower variables that may help in enhancing the efficiency of practicing audit environment, ranking score is 1, 75 and 1.77 respectively. Also, the non-governmental international organizations that influencing the audit practicing are classified as the lowest environmental variable in Egyptian practicing environment with a ranking score of (2.48).

The Second Component

The second component presents the loading of the Egyptian audit practicing environmental deficiencies and its associated inconvenience of audit environment that has the highest component loading score of 0.706, as well as, the auditor's limited responsibilities towards the audit services beneficiaries that ranked as the lowest loading score with 0.604.

The Third component

Third component presents the loading of the Egyptian audit practicing regulating deficiencies, such as outdated practicing law and regulations that has the highest component loading score of 0.720. While the lack of associated audit standards has the lowest loading score (0.604).

Regarding the External General Environment Correlation Analysis We analyzed the correlation between all of the suggested external environment variables in order to confirm our understanding of their interrelation effects. The following is a summary of analysis results:

- Weak positive coefficient correlation (0.492) between the efficiency of audit firms' followed procedures and non-governmental professional organizations that regulate audit practicing;
- Weak (0.301) positive coefficient correlation between practicing laws and both auditors' responsibilities and all other environment variables. The weakest coefficient correlation was for non-governmental professional organizations that regulate practicing;
- In relation to organizations that regulate and monitor audit practice, there was a strong positive correlation between the three main regulatory organizations. The co-efficient correlation between non-governmental professional organizations that regulate local practicing, and both governmental regulatory bodies and non-governmental international organizations were (.0606, and .0565), respectively, which indicates the importance of integrating these organizations' efforts during development of the practicing environment; typically, this can be achieved by increasing the role, authorities, and responsibilities of non-governmental professional organizations.
- There was a strong positive coefficient correlation between International Auditing Standards, International Financial Reporting Standards, and non-governmental international organizations, that regulate audit practicing, with a correlation factor of (0.578, and 0592), respectively, which confirms the strong integration between all of these variables, the importance of their parallel implementation, and the necessity of increasing cooperation between local and international professional non-governmental organizations.

9. Summary of Findings and Conclusion

Any organization is an open system that continuously interacts with its surrounding environmental variables; accordingly, congruence is required to harmonize the environmental internal and external variables and the dominant behaviors of the organization. Environmental changes do not usually require cultural changes. Culture changes involve a total restructuring of the environment, through changing the societal expectations (including regulation or deregulation). Due to the unique fundamentals and inefficiency of the Egyptian practicing local environment, and based on the empirical analysis results that were previously discussed, we may conclude that the Egyptian practicing local environment requires a deep culture change.

Some researchers in the field of management environment, such as **Gordon** (**1991**), suggests that changing the environment through imposing government actions does not always lead to the targeted outcomes. This assumption is not applicable for audit-practicing professions, as audit practicing is a key function in servicing society and protecting its interests, and accordingly it should be properly regulated.

Also, some environment early researches argued that close controls centralization through tightening laws and regulations will increase both governmental and non-governmental organizations' roles. Furthermore, this will decrease environmental uncertainty, and enhance the efficiency of the practicing environment.

In general, many organizational environments' researches concluded that organizations usually perform well, and properly understand customers' expectations, if they are working in a friendly, simple, and stable environment that offer substantial resources (Child, 1972; Cameron & Quinn, 2006). These high-performance organizations always have minimum environmental changes to deal with, thus they will be more focused on improving efficiency (Porter, 1985; Patrick & Maister, 2001). Based on the previously discussed literatures, we may conclude that the audit-practicing environment, in general, can be classified as a stable and complex environment, since there are many elements and variables included in this type of environment. However, these elements and variables are usually constant and change slowly. The recent financial scandals dramatically shifted the practicing environment's designation to be more complex, unstable, and uncertain. This shift resulted in many changes of law and regulations that govern practicing. **Pfeffer and Salancik (1978) and Brickley et al. (2003)** argue that greater uncertainty increases efforts to more centralization and coordination, which encourage formalizing of larger organizations through merger and acquisitions in order to face the increasing environment regulating steps and political control. The following Table presents the main characteristics of the Egyptian practicing local environment:

Environment	Reasons of Clas-	Local Environmental variables that Support the
Dimension	sification	Classification
Complex/	Degree to which	- Practicing environment has many variables that affect
State	the environment	it;
	variables remain	-Weak governing rules and regulatory bodies;
	basically the	-Unorganized and inefficient interrelationship between
	same over time,	governmental and nongovernmental organizations that
	or without con-	regulate and control the practicing decisions environ-
	tinual change;	ment; including unavailability of clear vision, mission
	-Ineffective role	or strategy for these organizations;
	of the variables;	-Poor practicing qualifications and rarity of human re-
	-Inefficient per-	sources;
	formance of the	-Outdated practicing law;
	variables;	-Inefficient relationship with international professional
	-Non-active in-	organizations;
	terrelation be-	-Improper development of local accounting and auditing
	tween the varia-	standards, which are not in line with similarly used in-
	bles	ternational standards;
		-Unavailability of local code of ethics or independency
		roles;
		-Improper interrelation between business industry laws
		and practicing regulations;
		-Unclear responsibilities of audit firms against stake-
		holders and community;
		-Unclear strategic goals or mission statement for audit-
		practicing firms.
Uncertainty	The trait of the	-No available clear structure of organizations that regu-
	environment re-	lates professional practicing;
	sulting from the	-No available authorized institutions to monitor quality
	two powerful	of services provided by practicing audit firms;

Characteristics of Egypt Practicing Local Environment

forces of com-	-No clear structures, or working procedures in most lo-
plexity and state	cal practicing firms;
	-Weak organizational culture of the most practicing au-
	dit firms;
	-Local professional practicing organizations are not
	generally following the global pattern or speed of de-
	velopment.

On the other hand, any environment changes require two levels (assumptions level and value level), therefore, empowering practicing organizations is necessary to meet both external and internal environment challenges. The successful implementations of these changes require modernizing of laws and regulations that organize audit practicing. They also require increasing the supported government and non-governmental governing bodies that regulate the practicing environment, in addition to creating proper strategic plans that provide effective communication between all environment key players through integrating their efforts to achieve the targeted overall development of the practicing environment.

In general, we extended our study research to understand the relationship between both the external and internal practicing environmental. We connected our measures to previous works in the field of the organizational environment, which provide us with the base to identify five practicing environmental constructs (variables) that consist of Non-Governmental Associations (Organizations), Audit Firms/Professional Organization, Governmental Regulatory Bodies (Organizations), Practicing Laws Auditors Characteristics, and International Associations. We identified the classification of Egypt's current practicing environment as dormant, complex, and uncertain. Also, we proposed that the local audit practicing environment is neither working efficiently nor effectively.

Based on our empirical analyses results and findings, there are a number of interesting observations related to the audit-practicing environment, services quality, and controls that may affect audit practicing in Egypt. The following table summarizes these observations, which are closely related to Egypt's local practicing environment's main weaknesses:

	Exists	Does not exist	To a certain ex- tent
a - Lack of auditing standards.	29%	41%	30%
b - Defects in some of the applied auditing standards.	33%	30%	37%
c- Outdated Audit Practicing Law and regula- tions.	61%	13%	26%
d - Auditor's limited responsibilities toward the beneficiaries.	42%	22%	37%
e- Easiness of obtaining practicing license.	56%	21%	24%
f - Inconvenience of auditing environment.	52%	12%	36%
g - The limited role of authorities regulating accounting and auditing.	64%	9%	27%

Empirical Analyses Observations Summary

Also, the following table indexes, in a ranking base, the different environment variables based on their magnitude on the audit-practicing environment:

Ranking Base of Practicing Environmental Variables

	Ranking Score
a- Laws and Procedures:	
1- The efficiency of the professional organization's procedures.	1.61
2- The law regulating audit practicing and the auditor's role.	1.39
b- Organizations:	
1- Non-governmental professional organizations that regulate auditing practice.	1.77
2- Monitoring governmental organizations that control audit practicing.	1.75
 Non-governmental international organizations regulating au- dit practicing. 	2.48
c- Standards:	
1- International auditing standards.	1.33
2- International financial reporting standards.	1.66

Summary of Findings and Conclusion

In summary, the empirical analyses results reveals that both local non-governmental and governmental organizations play a limited role in regulating Egypt's local practicing organizations compared to what is supposed to be as per international practicing standards. This limited role resulted in an inefficient and ineffective correlation between both the local practicing environment and practice professions, which is contrary to what is expected in a fully regulated practicing business. Local law regulating practicing does not present any interrelation, neither through regression nor correlation analysis, with any of the current weakness problems indicated in the local practicing environment, except for those related to the easiness of obtaining practicing licenses. These indicate the weakness and ineffectiveness of currently used practicing law, and confirm the immediate necessity of law reform procedures. The negative factors that are currently presented in the Egyptian local practicing environment led us to classify it as a complex and dormant environment with a limited degree of uncertainty. These types of environment need major reform procedures, mainly through increasing the role of both governmental and nongovernmental professional organizations.

The empirical findings, in general, confirmed that the Egyptian audit practicing law and professional regulatory bodies need to be reformed. Also, local professional auditors need to integrate their best practices in order to enhance their audit firms' practicing quality, and use procedures that monitor and evaluate the professional practicing process.

However, the magnitude of each variable in the practicing environment will be differentiated from one country to another based on the local culture and influence power of environment different variables. Accordingly, categorizing of environmental factors into groups for similar countries will be more efficient and save efforts during the study phase and implementation of audit harmonization.

Since Egypt is applying the Roman law system, any proposed practicing reforms require detailed laws to be issued by parliament to codify each type of accounting transactions, mainly if we consider the Egyptian basic values culture classification that has a low score of uncertainty avoidance and power distance.

Since the current Egyptian local practice environment is classified as a dormant,

complex, and uncertain environment that needs major reforms in order to increase its practice organizations' effectiveness. The Egyptian practice environment encourages the reduction of accounting and auditing profession's judgment role.

Also, empirical research results raised a consensus between the selected sample answers about the importance of increasing professional accountants and auditors' technical and judgmental capability, in addition to the necessity of increasing the non-governmental organizations' subscriptions in developing and managing the practicing profession.

10. Thesis Contribution and Motivation

The causes beyond the research results may be associated with the increase of globalization phenomena, and the influence of international audit-practicing environment on the way that the local Egyptian practice environment interacts. The main law governing the Egyptian professional accountants and auditors, particularly in the private sector, is the Accounting Practice Law 133/1951 and its amendments, which are considered out of date by most practitioners. The Company Law does not cover accounting and auditing standards, but requires that external audits be conducted in compliance with the Accounting Practice Law 133/1951. There is a gap between the Egyptian's law and the practice's compliance with accounting and auditing standards. Egyptian professional standards for accountants and auditors need to be improved by raising the quality of public university education and by introducing a professional qualifications examination.

The basic ground rules of the Egyptian political system have prevented necessary legislative reform. As such, this faulty legislative system has institutionalized a corrupt set of structures that led to biased decisions and occasionally outright corruption. Moreover, current local laws created an inefficient, unethical, and wasteful system. The auditor's independence in Egypt is strongly affected, which is aggravated by the lack of separation between audit and consultancy services provided by the auditor.

The thesis helps to develop the local audit practicing environment, which is expected to result in significant effects on the public auditing profession and practicing role. Also, it may create new balance for political forces, and increase the creditability of public audit services.

In general, profession reforms will support the auditor's position in the country's policy making through establishing a governmental relations module that enables the local standard-setting bodies to have a voice in issues affecting the local accounting professions. This can be achieved through the followings suggested Practicing General Reform Conceptual Framework:

Practicing General Reform Conceptual Framework

This framework focus on individual auditors development in order to support their professional judgment skills and capabilities to protect shareholders' interests through the fair presentation of financial information. Therefore, Egyptian regulators should also recognize the importance of applying IFAC Code of Ethics as the base for national independency codes of conduct.

Practicing General Reform Procedures

Developing of practicing performance quality involves changing the accounting and auditing statutory framework. to protect public interests through:

- Establishing of an accounting and auditing oversight body that focus on:
 - Preparing auditing standards for legislation;
 - Formulating auditing and assurance standards' guidance; and
 - Representing the local professions internationally.

The oversight body powers of authorities can be extended to:

- Engaging staff and consultants;
- Establishing committees and consultative groups; and
- Receiving money contributions to compensate its operating costs.
- Apply International Auditing and Accounting standards as a law.
- Develop proper monitoring and reporting mechanisms, including peer-review programs.
- Establish Competency Standards Committee to monitor the adequacy of professional and business ethics.
- Monitor auditors' annual independency declaration, and perform periodic independency tests to confirm that both practicing auditors and audit firms have the proper independency mechanism.

- Prohibit any kind of employment or financial relationships between auditors and their clients.
- Legalize the establishment of mandatory audit committees for listed companies
- Apply compulsory audit partners' rotation program,.
- Impose all publically listed companies to report their auditor's change to the Egyptian Stock Exchange
- Impose a Professional Code of Ethics for all auditors
- Apply licensing system that meets IFAC requirements
- Implement maximum civil penalty for audit misleading or fraud.
- Oblige chief executive officer and chief financial officer of audited companies to sign the company's audited financial statements.
- Impose auditors to form an opinion on the company's corporate governance standards.
- Prohibit auditors from providing their audited clients with non-audit services, mainly for joint stock companies, except in cases approved by the clients' audit committee.
- Enhancing the Role of Other Regulatory Agencies/Ministry of Finance, as the ministry of finance (MOF) should have more powerful controlling and monitoring authorities in order to properly act on behalf of the government, therefore, it should have the right to:
 - Give a professional accounting body ESAA, "Egyptian Society for Accountants and Auditors," a written notice to request a copy of the society's code of professional conduct, and any related amendments.
 - Request any local audit firm or individual auditor to provide information about any completed audit assignment,
 - Punish and penalize any audit firm, or individual auditor, that commits any non co-piling auditing or accounting standards offence.
- Developing the Quality of Audit-Practicing Organizations through :

- Developing the Role of Egyptian Society of Accountants and Auditors (ESAA) to include :
 - Regularly examine the appropriateness of overseeing public interests mechanism, supporting continuous education, and training quality of accountants and auditors
 - Monitor the recent changes on the international accounting and auditing standards, and evaluate their effects on the local profession.
 - Review of both auditors and audit firms' compliance standards, and impose actions against violators.
 - Assess the effectiveness of local audit firms' operating systems to ensure their independency compliance. Measure the adequacy of auditors' local independency rules, Audit Practicing Act, and Codes of Professional Conduct.
 - Perform periodic quality-assurance reviews for the local audit firms.
 - Monitor companies' compliance procedures in relation to disclosures' arrangement rules and regulations.
 - Provide its members with a future clear vision of the professions that help them to shape their practicing development plan throughout their careers.
 - Provide guidance of expected competencies for newly qualified professional auditors.
 - Support Ministry of Finance activities through providing it with a yearly report that discusses the society performance, and any actions taken against any individual auditor or audit firm.
- Improving Audit Firms' Quality Management Through enhancing :
 - Internal Leadership Responsibilities for Quality Within the Firm
 - Applying of Ethical Requirements

- Acceptance and Continuation of Clients' Relationships and Specific Engagements
- Human Resources Management Procedures
- Monitoring and control procedures

- Abbott, A. (1988). *The system of the professions: An essay on the division of expert labor*. Chicago: University of Chicago Press.
- Abd-Elsalam, O.H., & Weetman, P. (2003). Introducing international accounting standards to an emerging capital market: relative familiarity and language effect in Egypt. *Journal of International Accounting, Auditing & Taxation*, 12, 63–84.
- Ackroyd, S. & Thompson, P. (1999). Organizational misbehavior. London: Sage.
- Agrawal, A., & Chadha, S. (2005). Corporate governance and accounting scandals. *Journal of Law and Economics*, (XLVIII), 371–378.
- AICPA. (2003). Statements on quality control standards. *American Institute of Certified Public Accountants*, New York, 2–10.
- AICPA. (2009). Standards for performing and reporting on peer reviews. American Institute of Certified Public Accountants, New York, 1006–1020.
- Akhigbe, A., Martin, A., & Nishikawa, T. (2009). Changes in risk of foreign firms listed in the U.S. following Sarbanes-Oxley. *Journal of Multinational Financial Management*, (19), 195–205.
- Aldrich, H. (1979). Organizations and environments. NJ: Englewood Cliffs Prentice-Hall.
- Aldrich, H.E., Ruef, M. (2nd ed.). (2006). Organizations evolving. London: Sage.
- Allaire, Y., & Firsirotu, M. E. (1984). Theories of organizational culture. *Organization Studies*, 5, 193–226.
- Allen, W. A. & Moessner, R. (2011, June). The international liquidity crisis of 2008–2009. *World Economic*, 12(2).

- American Management Association (AMA). (2006). The ethical enterprise doing the right things in right ways, today and tomorrow: A global study of business ethics 2005–2015. *Human Resource Institute*, 2–16.
- Arens, A. A., Elder, R.J. & Beasley, M. S. (2005). *Auditing and assurance services: an integrated approach*. New Jersey, Pearson: Prentice Hall.
- Armstrong, M. B. (2002). Ethical issues in accounting. The Blackwell guide to business ethics. Oxford: Blackwell Publishing.
- Arnold, P. (2009). Global financial crisis: the challenge to accounting research. *Accounting Organizations and Society*, 34(6/7), 803–809.
- Arya, A., Glover, J.C. & Sunder, S. (2003). Are unmanaged earnings always better for shareholders?. *Accounting Horizons*, 17, 111–116.
- Australian Accounting Research Foundation. (2002). Corporate law economic reform program: corporate disclosure strengthening the financial reporting framework. *Legislation Review Board Position Paper*, 4–20.
- Baier, A. (1986). Trust and antitrust. *Journal of Ethics, The University of Chicago Press*, 96(2), 259–260.
- Baker, C. R. (2002). Investigating Enron as a public private partnership. *Accounting, Auditing & Accountability Journal*, 16(3), 1–20.
- Baker, C. R., Bédard, J. & Prat Dit Hauret, C. (2005, May). A comparative analysis of regulation of statutory auditing in France, Canada and the United States. *European Accounting Association Congress*.
- Ball, R. (2009, May). Market and political/regulatory perspectives on the recent accounting scandals. *Journal of Accounting Research*, 47, 1–53.
- Ball, R., Kothari, S., & Robin, A. (2000). The effect of international institutional factors on properties of accounting earnings. *Journal of Accounting and Economics*, 29, 1–51.

- Barth, J. R., Caprio, G. Jr., & Levine, R. (2006). *Rethinking bank regulation: till angels govern*. New York: Cambridge University Press.
- Barth, M., Landsman, W. & Lang, M. (2008). International accounting standards and accounting quality. *Journal of Accounting Research*, 46(3), 467–498.
- Bastedo, M. N. (2004). Open systems theory. *The SAGE Encyclopedia of Educational Leadership and Administration, University of Michigan*, 1–2.
- Beaulieu, P. & Reinstein, A. (2006). The effect of accounting research on the beliefs of practitioners: The case of auditor independence. *Auditing Midyear Conference*, CA.
- Beck N., Brüderl J., & Woywode, M. (2008). Momentum or deceleration? Theoretical and methodological reflections on the analysis of organizational change. *Academy of Management Journal*, 51(3), 413–435.
- Behn, B. K., Carcello, J.V., Hermanson, D.R., & Hermanson, R. H. (1999). Client satisfaction and big 6 audit fees. *Contemporary Accounting Research*, 16(4), 587– 608.
- Benston, G., Bromwich, M., & Wagenhofer, A. (2006). Principles-versus rules-based accounting standards: The FASB's standard setting strategy. *Abacus*, 42(2), 165–188.
- Bogle, J. C. (2000, October 16). Public accounting: profession or business? The Seymour Jones Distinguished Lecture. *Vincent C. Ross Institute of Accounting Research*, New York University, 3–7.
- Bonner, S.E., Palmrose, Z., & Young, S. M. (1998, October). Fraud type and auditor litigation: An analysis of SEC accounting and auditing enforcement releases. *The Accounting Review*, 73(4), 503–532.
- Bourgeois, L. (1980). Strategy and environment: a conceptual integration. Academy of Management Review, 5, 25–35.

- Boyd, B., Dess, G., & Rasheed, A. (1993). Divergence between archival and perceptual measures of the environment: causes and consequences. *Academy of Management Review*, (18), 204–215.
- Boyd, C. (2004). The structural origins of conflict of interest in the accounting profession. *Business Ethics Quarterly*, 14(3), 377–398.
- Brandon, D. M., & J. M. Mueller. (2006). The influence of client importance on juror evaluations of auditor liability. *Behavioral Research in Accounting*, 18, 1–18.
- Brickley, J. A., Smith, C. W., & Zimmerman, J. L. (2003). *Designing organizations to create value: from structure to strategy*. New York, NY: McGraw-Hill.
- Burns, T., & Stalker, G. M. (1961). The management of innovation. London: Tavistock.
- Butler, H. N., & Ribstein, L. E. (2006). *The Sarbanes-Oxley debacle: What we've learned; How To Fix It.* Washington, DC: AEI Press.
- Cameron, K. S., & Quinn, R. E. (2006). *Diagnosing and changing organizational Culture based on the competing values framework*. San Francisco, CA: John Wiley & Sons, Inc.
- Caneghem, V. T. (2004). The impact of audit quality on earnings rounding-up behavior: Some UK evidences. *European Accounting Review*, 13(4),771–786.
- Caprio, G. Jr. (2009). Financial regulation in a changing world: lessons from the recent crisis. *Journal of Financial Stability*, 13, 205–248.
- Carmichael, S., Hummels, H., Klooster, A., & Luijk, V. H. (2008). How ethical auditing can help companies compete more effectively at the international level. 2–8. Retrieved from: http://www.nd.edu/~isbee/ papers.htm.
- Catanach, A. H., & Catanach, S. R. (2003). Enron: a financial Reporting Failure? *Villanova Law Review*, 48(4), 1057–1065.

- Chandler, R. & Edwards, J. R. (1996). Recurring issues in auditing: back to the future? *Cardiff Business School Accounting, Auditing & Accountability Journal*, 9(2), 4–29.
- Chang, H., Choy, H., Cooper, W., & Ruefli, T. (2009). Using Malmquist Indexes to measure changes in the productivity and efficiency of U.S. accounting firms before and after the Sarbanes-Oxley Act. *Omega*, 37(5), 951–960. Retrieved from: http://www.sciencedirect.com/science/article/B6VC4242c515ed803c28eb91a
- Chenhall, R.H. (2006). Theorizing contingencies in management control systems research. *Handbook of Management Accounting Research*, 1, 165–175.
- Child, J. (1972). Organizational structure, environment and performance: The role of strategic choice: The specialist strategy. *California Management Review*, 26(3), 2–22.
- Choi, F.D.S., & Meek, G.K. (2005). *International accounting*. New Jersey: Prentice Hall.
- Choi, J. H., Doogar, R., & Ganguly, A. R. (2004). The riskiness of large audit firm client portfolios and changes in audit liability regimes: evidence from the U.S. audit market. *Contemporary Accounting Research*, 21, 747–785.
- CICA. (2005). The CA competency map: the pervasive qualities professional skills. *Canadian Institute of Chartered Accountants*, 2–22.
- Coffee, J. C., & Berle, A. (2001). The acquiescent gatekeeper: reputational intermediaries, auditor independence and the governance of accounting. *Social Science Research Network, Columbia University Law School*, 6–51. Retrieved from: http://papers.ssrn.com/paper.taf?abstract_id=270944
- Conroy, S. J., & Emerson, T. L. N. (2006, June). Changing ethical attitudes: The case of the Enron and In Clone scandals. *Southwestern Social Science Association*, 87(2), 397–398.

- Cooper, D., & Robson, K. (2006). Accounting professions and regulation: locating the sites of professionalization. Accounting Organizations and Society, 31(4/5), 415–444.
- Cunningham, L. A. (2004). Facilitating auditing's new early warning system: control disclosure, auditor liability, and safe Harbors. *Boston College Law School*, 4–27.
- Dahawy, K., Merino, B.D., & Conover, T.L. (2002). The conflict between IAS disclosure requirements and the secretive culture in Egypt. *Advances in International Accounting*, 15, 203–228.
- Danielle, M. (2003). Controllers or catalysts for change and improvement: Would the real value for money auditors please stand up? *Ecole des Hautes Etudes Commerciales, Managerial Auditing Journal*, 8–29.
- Davis, S. M. (1984). Managing corporate culture. Cambridge: MA Ballinger.
- Dechow, P., & Skinner, D. (2000). Earnings management: Reconciling the views of accounting academics, practitioners, and regulators. *Accounting Horizons*, 14, 235–250.
- Deegan, C., & Unerman, J. (3rd. ed.). (2009). *Financial accounting theory*. McGraw-Hill.
- DeFond, M. L. & Francis, J. R. (2005). Audit research after Sarbanes-Oxley. *Journal of Practice and Theory*, 24, 5–30.
- DeGeorge, R. (5th. ed.). (1999). Professions in business and professions as business: Business Ethics. New Jersey: Prentice-Hall.
- Dharan, B. G. (2002). Business education and corporate accounting scandals: lessons on accounting information and investor. *Brazos Higher Education Service Corp*, 918–925.

- Dimaggio, P. J., & Walter W. P. (1983). The iron cage revisited: Institutional isomorphism and collective rationality in organizational fields. *American Sociological Review*, 48, 150-158. Retrieved from: www.cica.ca/becoming-a ca/documents/item9604.pdf
- Domingo, G. (2005). Trust and Dialogue: Theoretical approaches to ethics auditing. *Journal of Business Ethics*, 57, 209–219.
- Draghi, M. (2008, November 13). A Vision of a more resilient global economy. *Financial Times*. Retrieved from: http://www.ft.com/cms/s/0/4dbe4b1a-b18f-11ddb97a-0000779fd18c.html#axzz2hZD1puYx
- Duncan. R. B. (1972). Characteristics of organizational environments and perceived environmental uncertainty. *Administrative Science Quarterly*, 17(3), 314–320.
- Duska, R. F. (2004). Six cures for current ethical breakdowns. *Journal of Financial Service Professionals*, 58(3), 23–37.
- Duska, R. F. (2005). The good auditor skeptic or wealth accumulator? Ethical lessons learned from the Arthur Andersen debacle. *Journal of Business Ethics*, 57, 17–29.
- Duska, R.F., Duska, B. S., & Ragatz, J. (2nd ed.). (2011). Accounting ethics. John Wiley & Sons, Ltd.
- Eisenberg, T., & Macey, J. R. (2004, July). Was Arthur Andersen different? An empirical examination of major accounting firm audits of large clients. *Journal of Empirical Legal Studies*, 1(2), 263–298.
- El-Din, Z. B., (2005). Egypt code of corporate governance guidelines and standards. *MENA-OECD Investment Programme:* Egypt, 10–15.
- Epstein, M. J., & Geiger, M. A. (1994). Investor views of auditor assurance: recent evidence of the expectation gap. *Journal of Accountancy*, 177(1), 60–64.
- Estafen. B. D. (1971). Methods for management research in the 1970's: An ecological systems approach. *Academy of Management Journal*, 14, 51–64.

- Ettredge, M. L., Heintz, J., Li, C., & Scholz, S. (2011). Auditor realignments accompanying implementation of SOX 404 ICFR reporting requirements. *American Accounting Association*, 18–25. Retrieved from: http://ssrn.com/abstract=874836.
- Fawzy, S. 2003. Assessment of corporate governance in Egypt. Egyptian Center Economic Studies, 1–15. Retrieved from: http://www.eiod.org/uploads/Publications/Pdf/assessingcgegypt.pdf.
- Fernandez, S., & Rainey, H. (2006). Managing successful organizational change in the public sector. *Public Administration Review*, 66, 168–176.
- Fields, T., Lys, T., Vincent, L. (2001). Empirical research on accounting choice. *Journal of Accounting and Economics*, 31, 260–270.
- Flegm, E. (2005). Accounting at a crossroad. *The CPA Journal*, New York State Society of CPAs, 2–8. Retrieved from: http://www.nysscpa.org/cpajournal/2005/1205/infocus/p16.htm
- Fox, Loren. (2003). *Enron: the rise and fall*. Hoboken, New Jersey: John Wiley and Sons.
- Francis, J., & Yu, M. (2009, September). The effect of big four offices size on audit quality. *The Accounting Review*, 84, 1521–1552.
- Frankel, R., Johnson, M., & Nelson, K. (2002). The relation between auditors' fees for Non-audit services and earnings management. *The Accounting Review*, 77, 71–105.
- FSF's. (2008, April). Enhancing market and institutional resilience. *Financial Stability Board Publications*, 1–14. Retrieved from: http://www.financialstabilityboard.org/publications/ r_0804.pdf.
- Gaa J. C., & Thorne, L. (2004). An Introduction to the special issue on professionalism and ethics in accounting education, *Issues in Accounting Education*, 19 (1), 1– 6.
- Gagliardi, P. (1986). The creation and change of organizational cultures: A conceptual framework. *Organization Studies*, 7(2), 117–134.

- GAO, U.S. Government Accountability Office. (2003, November 13). The future of the accounting profession: American Assembly Report, 103 rd American Assembly. *The American Assembly, Columbia University*. Retrieved from: http://www.gao.gov/cghome/amacas/text1.html
- Garg, V. K., Walters, B. A., & Priem, R. L. (2003). Chief executive scanning emphases, environmental dynamism, and manufacturing company performance. *Strategic Management Journal*, 24, 727–734.
- Garry, D. C., & Napier, C. J. (2002). Exploring comparative international accounting history. *Accounting, Auditing and Accountability Journal*, 15(5), 690–702.
- Gary, R. N., Owen, D., Evans, R., & Zadek, S. (1997). Struggling with the praxis of social accounting: shareholders, accountability, audits and procedures. Center of social and environmental accounting research, university of Dundee, 2-14.
- Geiger, M. A., & Raghunandan, K. (2002). Conservatism of the big six audit firms and going concern modified audit reports. *Academy of Accounting and Financial Studies Journal*, 6(1), 1–7.
- Gherai, D. S., & Balaciu, D. E. (2011). From creative accounting practices and Enron phenomenon to the current financial crisis. *Annales Universitatis Apulensis Series Oeconomica*, 13(1), 34–39.
- Gilbert, C.G. (2005). Unbundling the structure of inertia: Resource versus routine rigidity. *Academy of Management Journal*, 48(5), 741–763.
- Gilbert, C.G. (2006). Change in the presence of residual fit: Can competing frames coexist? *Organization Science*, 17(1), 150–167.
- Gordon, J. N. (2003, March). Governance failures of the Enron board and the new information order of Sarbanes-Oxley. Working paper. *Columbia University, Law School Review Symposium*, 3–16.
- Gordon, S., Stewart, W., Sweo, R., & Luker, W. (2000). Convergence versus strategic reorientation: The antecedents of fast-paced organizational change. *Journal of Management*, 26, 915–933.

- Government Accounting Office (GAO). (2003). Public accounting firms: mandated study on consolidation and competition. United States General Accounting Office Report to the Senate Committee on Banking, Housing, and Urban Affairs and the House Committee on Financial Services, 1158.
- Gramling, A. A., & Stone, D. N. (2001). Audit firm industry expertise: A review and synthesis of the archival literature. *The Accounting Review*, 74(2), 201–216.
- Gray, S. J. (1988). Towards a theory of cultural influence on the development of accounting systems internationally. *ABACUS*, 24(1), 1–15.
- Greenwood, R., & Lachman, R. (1996). Change as an underlying theme in professional service organizations: An introduction. *European Group for Organizational Studies*, 17: 5–8.
- Gul, F. A. (1999). Audit prices, product differentiation and economic equilibrium. auditing. *Journal of Practice and Theory*, 30(2), 92–98.
- Gunther, J. W. & Moore, R. R. (2002). Auditing the auditors: Oversight or overkill? *Economic & Financial Policy Review, Federal Reserve Bank of Dallas*, 1(5), 4–15.
- Gupta, P. P., Dirsmith, M. W., & Fogarty, T. J. (1994). Coordination and control in a government agency: Contingency and institutional theory perspectives on GAO audits. *Administrative Science Quarterly*, 39(2), 363.
- Harris, R. (2004). Organizational task environments: An evaluation of convergent and discriminate validity. *Journal of Management Studies*, 41, 859–862.
- Hartmann, F. G. H. (2000). The appropriateness of RAPM: Toward the further development of theory. *Accounting Organizations and Society*, 25 (4–5), 453–465.
- Hegarty, J., Gielen, F., & Hirata-Barros, A. C. (2004, September). Implementation of international accounting and auditing standards: Lessons learned from the World Bank's. Accounting and Auditing ROSC Program, 13:15. Retrieved from: http://www-wds.worldbank.org/external/default/WDSContent-Server/WDSP/IB/2008/01/22/000333037_20080122014815/Rendered/PDF/419530WP0Lesso1ned1ROSC1AA01PUBLIC1.pdf

- Hetzel, R. (2009). Should increased regulation of bank risk-taking come from regulators or from the market? *Federal Reserve Bank of Richmond Journal Economic Quarterly Review*, 95, 161–200.
- Hill, C. A. (2004). Regulating the rating agencies. *Washington University Law Quarterly*, 43, 43.
- Hillman, A. C. (2005). Politicians on the board: do connections affect the bottom line? *Journal of Management*, 31(3), 464–481. doi: 10.1177/0149206304272187.
- Hofstede, G. (1980). Culture's consequences: International differences in work-related values. *Society for Applied Anthropology*, 41(1), 86–90.
- Hofstede, G. (1991). *Culture and organizations: Software of the mind*. London: McGraw-Hill.
- Hoitash, A. M., & Charles A. B. (2005). Auditor fees, abnormal fees and audit quality before and after the Sarbanes Oxley Act. *Social Science Electronic Publishing*, 3–10. Retrieved from: http://papers.ssrn.com/sol3/papers.cfm?abstract_id=646681
- Holm, C., & Laursen, P.B. (2007). Risk and control developments in corporate governance: changing the role of the external auditor. *Corporate Governance Magazine*, 15(2), 322.
- Holmstrom, B., & Kaplan, S. N. (2003). The state of U.S. corporate governance: What's right and what's wrong? *Journal of Applied Corporate Finance*, 15(3), 8–20.
- Hope, O., (2003). Firm-level disclosures and the relative roles of culture and legal origin. *Journal of International Financial Management and Accounting*. 14(3), 218–248.
- Humphrey, C., Loft, A., & Woods, M. (2009). The Global audit profession and the international financial architecture: Understanding regulatory relationships at a time of financial crisis. *Accounting Organizations and Society*, 34(6/7), 810– 825.

- Hutzschenreuter, T., & Kleindienst, I. (2006). Strategy-Process research: What have we learned and what is still to be explored. *Journal of Management*, 32(5), 676–712. doi: 10.1177/0149206306291485
- ICAA. (1998). The vision 2020 taskforce report chartered accountancy in the next century: Radical change or diminished influence? *Institute of Chartered Accountants in Australia, Chartered Accountants House*, 3–7.
- ICAA. (2005). Quality control for accounting firms. *The Institute of CPA Australia*, 32–33. Retrieved from: http://www.icaa.org.au/upload/download/APS5_Survey.pdf.
- IFAC. (2003). Rebuilding public confidence in financial reporting: An international perspective. *International Federation of Accountants*, 1–38. Retrieved from: http://www.ifac.org/news-events?qnode/view/73.
- Institute of Chartered Accountants in Australia. (2004, November). Auditing competency standard for registered company auditors. *Chartered Accountants House (ICAA)*, 1–6.
- International Organization of Securities Commissions (IOSCO). (2008, December 11). Technical Committee of IOSCO Code of Conduct Fundamentals for CRAs. *Technical Committee of the International Organization of Securities Commissions*, 2–16. Retrieved from: http://www.iosco.org/library/pubdocs/pdf/IOSCOPD270.pdf.
- IOSCO, The International Organization of Securities Commissions. IASC 2000 Standards. Retrieved from: http://www.iosco.org/press/presscomm000517.html.
- Jaggi, B., & Low, P.Y. (2000). Impact of culture, market forces, and legal system on financial disclosures. *International Journal of Accounting*, 35(4), 495–519.
- Jamal, K., & Tan, H. T. (2010). Joint effects of principles-based versus rules-based standards and auditor type in constraining financial managers' aggressive reporting. *The Accounting Review*, 85(4), 1325–1346.

- Jaskyte, K., & Dressler, W. (2004, August). Studying culture as an integral aggregate variable: organizational culture and innovation in a group of non-profit organizations. *Field Methods*, 14(3), 265–284.
- Javidan, M., Dorfman, P. W., Deluque, M. S., and House, R. J. 2006. In the eye of the beholder: cross cultural lessons in leadership from project GLOBE. Academy of Management Perspectives, 67–90.Gordon, G. (1991). Industry determinants of organizational culture. Academy of Management Review, 16(2): 396– 415.
- Jeffreys, D. Practice, A. (2002). New assurance standards on quality control. *Ontario Chartered of Accountants*, 1. Retrieved from: http://www.cica.ca/publications/list-of-publications/item60942.aspx
- Johnson, W.B., & Lys, T. (1990). The market for audit services: evidence from voluntary auditor changes. *Journal of Accounting and Economics*, 12, 281–308.
- Jones, G. R. (5th ed.). (2007). Organizational theory design, and change. Pearson Education.
- Jones, T. W., & Bowie, N. E. (1998). Moral hazards on the road to the 'virtual' corporation. *Business Ethics Review*, 8(2), 275–290.
- Kahneman, D., & Tversky, A. (2000). *Choices values and frames*. New York: Cambridge University Press.
- Keats, B. W., & Hitt, M. A. (1988). Linkages among environmental dimensions and macro-organizational characteristics: A causal modeling approach. *Academy of Management Review*, 31(3), 570–598.
- Kedia, S., & Philippon, T. (2005). The economics of fraudulent accounting. *Quarterly Journal of Economics*, 22(6), 655–669.
- Kedia, S., Koh, K., & Rajgopal, S., (2010). Evidence on contagion in corporate misconduct. Social Science Electronic Publishing, 1–12. Retrieved from: http://papers.ssrn.com/sol3/papers.cfm?abstract_id=1571538

- Klaus, P. (2008). The hard truth about soft skills workplace lessons smart people wish they'd learned sooner. Harper Collins.
- Kotter, J. P., & Heskett, J. L. (1992). *Corporate culture and performance*. New York: The Free Press.
- Kreiser, P., & Marino, L. (2002). Analyzing the historical development of the environmental uncertainty construct. *Management Decision*, 40(9), 896–900. doi:10.1108/00251740210441090
- Krishnamurthy, S., Zhou, J., & Zhou, N. (2006). Auditor reputation, auditor independence, and the stock market reaction to Andersen's Clients. *Contemporary Accounting Research*, 23(2), 1–6.
- Krishnan, L. (2011). *Legal issues on the scandals involving auditors*. 2nd. International Conference on Business and Economic Research. 1315–1325. Retrieved from: http://www.internationalconference.com.my/proceeding/2ndicer2011_proceeding/2552nd%20ICBER%202011%20PG%20135-1329%20Scandals%20Involving%20Auditors.pdf
- Largay, J. (2002). Lessons from Enron. Accounting Horizons, 16(2), 154.
- Larsen E. R., & Lomi, A.(2002). Representing change: a system model of organizational inertia and capabilities as dynamic accumulation process. *Simulation Modeling Practice and Theory*, 10(5/7), 274–286.
- Lawrence, P., & Jay L. (1967). Organization and environment; managing differentiation and integration. Homewood, IL: R.D. Irwin.
- Leifer, R., & Huber, G. P. (1977). Relations among perceived environmental uncertainty organization structure and boundary spanning. *Administrative Science Quarterly*, 22(2), 240–244.
- Lenz, H. (2008). The rational auditor and moral norms: The code of ethics for professional accountants lip service or self-binding agreement? University Wuerzburg, Germany, 1:18. Retrieved from: https://aaahq.org/meetings/AUD2009/TheRationalAuditor.pdf

- Leung, P., & Cooper, B. J. (2005). Accountants ethical issues and the corporate governance context. *Australian Accounting Review*, 15(1), 1–3.
- Leuz, C., Nanda, D., & Wysocki, P.D. (2003). Earnings management and investor protection: an international comparison. *Journal of Financial Economics*, 69(3), 505–27.
- Libby, T., & Thorne, L. (2004). The identification and categorization of auditors' virtues. *Business Ethics Quarterly*, 14(3), 479–498.
- Lovell, A. (1995). Moral reasoning and moral atmosphere in the domain of accounting. *Accounting Auditing & Accountability Journal*, 8(3), 60–75.
- Low, M., Davey, H., Hooper, K. (2008). Accounting scandals, ethical dilemmas and educational changes. *Elsevier*, 19(2), 222–254.
- Lynn, L., Heinrich, C., & Hill, C. (2000). Studying governance and public management: Challenges and prospects. *Journal of Public Administration Research and Theory*, 10(2), 233–261.
- Macnamara, D., & Banff, E. (2004). *Improving governance performance rules-based vs principles-based approaches*. From Leadership Acumen Doug Macnamara & Banff Executive Leadership, 16,1:5. Retrieved from: http://www.banffex-eclead.com/AcumenPDF/Governance%20Articles/Leadership%20Acu-men%2016%20V10%20Governance%20Principles%20vs%20Rules.pdf
- Mahoney, P. G. (2009). The development of securities law in the United States. *Journal* of Accounting Research, 47(2), 325–347.
- March, J., & Johan O. (1989). *Rediscovering institutions: The organizational basis of politics*. New York: Free Press.
- Mark. D. R. (1995). Business ethics, where do we stand? Towards a new inquiry. *Management Decision*, 34(4), 54–61.
- Marshall, J. (2009). The financial crisis in the US: Key events, causes and responses. Business and Transport Section, House of Commons Library: UK, 9(34), 7– 10.

- McLean, <u>B.,</u> & Elkind, <u>P</u>. (2003). *The smartest guys in the room: The amazing rise and scandalous fall of Enron*. London: Penguin Group.
- McMullen, J. S. & Shepherd, D. A. (2006). Entrepreneurial action and the role of uncertainty in the theory of the entrepreneur. *Academy of Management Review*, 31(1), 132–144.
- Meyer, J., & Rowan, B. (1977). Institutionalized organizations: Formal structure as myth and ceremony in the new institutionalism in organizational analysis. American Journal of Sociology, 83(2), 340–363.
- Miller, D. 1987. The structural and environmental correlates of business strategy. *Strategic Management. Industrial Relations*, 8(1), 55–76.
- Miller, K. D. (1993). Industry and country effects on manager's perceptions of environmental uncertainties. *Journal of International Business Studies*, 24(3), 693– 705.
- Miner, J. B. (1981). Scoring guide for the miner sentence completion scale form. NY: Organizational Measurement Systems Press.
- Miner, J. B., Crane, D. P., & Vandenberg, R. J. (1994). Congruence and fit in professional role motivation theory. *Organization Science Review*, 5(1), 89–93.
- Mintzberg, H. (1979). *The structuring of organizations*. NJ: Englewood Cliffs Prentice-Hall.
- Mitchell, L. E. (2003). The Sarbanes-Oxley Act and the reinvention of corporate governance. *Villanova Law Review*, 48 (4), 1190–1199.
- Moon, M. J. (1999). The pursuit of managerial entrepreneurship: Does organization matter? *Public Administration Review*, 59(1), 31–43.
- Moore, D. A, Cain, D. M., & Loewenstein, G. (2005). The dirt on coming clean: perverse effects of disclosing conflicts of interest. *Journal of Legal Studies*, 34, 1–8.

- Moore, D. A., Tanlu L., & Bazerman, M. H. (2010). Conflict of interest, and the unconscious intrusion of bias. *Judgment and Decision Making*, 5(1), 37–46.
- Moore, D. A., Tetlock, P. E., Tanlu, L.,& Bazerman, M. H. (2006). Conflicts of interest and the case of auditor independence: Moral seduction and strategic issue cycling. *Academy of Management Review*, 31(1), 12–18.
- Moore, G.A. (2000). Living on the fault line. New York: Harper Collins.
- Mulford, C., & E. Comiskey. (2002). *The financial numbers game: Detecting creative accounting practices*. New York: John Wiley & Sons.
- Myers, J.N. Myers, L. A. and Omer, T.C. (2003). Exploring the term of the auditor-client relationship and the quality of earnings: a case for mandatory auditor rotation? *Accounting Review*, 78 (3), 779–99.
- National Association of Certified Fraud Examiners (ACFE). (1993). Cooking the books: What every accountant should know about fraud. *NACFE Self-Study Workbook, No.* 92(5401),12.
- Nelson, M.W., Elliott, J.A. & Tarpley, R.L. (2003). How are earnings managed? Examples from auditors. *Accounting Horizons supplement*, 17–35. Retrieved from: http://fisher.osu.edu/~young.53/Nelson-Elliott-Tarpley.pdf
- Nicholas, C. M., & Neil, R. L. (1995). A socio-economic paradigm for analyzing managers' accounting choice behavior. *Flinders University of South Australia Accounting, Auditing & Accountability Journal*, 8 (1), 38–62.
- Nicolai, J. F., & Laursen, K. (2005). Performance pay, delegation and multitasking under uncertainty and innovativeness: An empirical investigation. *Journal of Economic Behavior & Organization*, 58 (2), 246–276.
- Nobes, C., & Parker, R. (10th ed.). (2008). *Comparative international accounting*. Prentice Hall.
- Nobes, C., & Parker, R. (2004). *Comparative international accounting*. Harlow: Pearson Education.

- Olson, M. (1965). *The logic of collective action*. Cambridge, MA: Harvard University Press.
- Palmrose, Z. (1988). An analysis of auditor litigation and audit service quality. *The Accounting Review*, 64(1), 55–73.
- Palmrose, Z., & Scholz, S. (2004). The circumstances and legal consequences of non-GAAP reporting: evidence from restatements. *Contemporary Accounting Research*, 21(1), 139–140.
- Palmrose, Z.V., Richardson, V.J., & Scholz, S. (2001). Determinants of market reactions to restatement announcements. *Journal of Accounting and Economics*, 37, 59–90.
- Pandit, G.M. (1999). Clients' perceptions of their incumbent auditors and their loyalty to the audit firms: an empirical study. *Mid-Atlantic Journal of Business*, 35(4), 171–88.
- Parker, R.H., & Nobes, C.W. (1994). *An international view of true and fair accounting*. London: Routledge, Edward Elgar Publishing.
- Patrick, J. M., & Maister, D. H. (2001). *First among equals how to manage a group of professionals*. NY: The free press.
- Pendergast, M. A. (2005). Harmonization of independence standards for accountants around the world. *IFAC Ethics Committee Activity Report*, 2–7. Retrieved from: http://www.ifac.org/Library/SpeechArticle.tmpl?NID=99133216218727
- Pennings, J. (1992). Structural contingency theory: A reappraisal. *Research in Organizational Behavior*, 14, 267–309.
- Perera, H. (1989). Towards a framework to analyze the impact of culture on accounting. *The International Journal of Accounting*, 24(1), 42–56.
- Perry, J. L., & Lois R. W. (1990). The motivational bases of public service. *Public Administration Review*, 50(3), 3–6.
- Perry, L. G. (1985). The regulation of the accounting profession and the problem of enforcement. *Journal of Comparative Business and Capital Market Law*, 7, 292:303.
- Pfeffer, J., & Salancik, G. R. (1978). *The external control of organizations: A resource dependence perspective*. New York: Harper & Row.
- Piotroski, J., & Srinivasan, S. (2008). Regulation and bonding: The Sarbanes-Oxley Act and the flow of international listings. *Journal of Accounting Research*, 46(2), 383–425.
- Pooran, P. N. (2010). Global financial crisis: A mid-year review of progress to date. *World Economics*, 11(3), 2:5
- Porter, B. A. (1996). A research study on financial reporting and auditing bridging the expectation gap. *Accounting Horizons*, 10(1),130–135.
- Porter, B., Simon, J., & Hatherly, D. (2005). *Principles of external auditing*. John Wiley & Sons.
- Porter, M.E. (1985). Competitive advantage. New York.
- Power, B., & Reid, G. (2005). Flexibility, firm-specific turbulence and the performance of the long-lived small firm. *Review of Industrial Organization*, 26(4), 416–426.
- Prajogo, D. (2006). The relationship between innovation and business performance: a comparative study between manufacturing and service firms. *Knowledge and Process Management*, 13(3), 220–225.
- Prendergast, C. (2002). The tenuous tradeoff between risk and incentives. *Journal of Political Economy*, 110(5), 1071–1102.
- Previts, G. J. (2005, October). Seven voices from the profession. *Journal of Accountancy Centennial Issue*, 1–5. Retrieved from: http://clubs.cob.calpoly.edu/~cmiller/ARTICLES/Background/7%20voices%20from%20the%20profession.doc

- Priem, R. L., Love, L. G. & Shaffer, M. A. (2002). Executives' perceptions of uncertainty sources: a numerical taxonomy and underlying dimensions. *Journal of Management*, 28(6), 730–742.
- Priem, R., Rasheed, A., & Kotulic, A. (1995). Rationality in strategic decision processes, environmental dynamism and firm performance. *Journal of Management*, 21(5), 913–929.
- Putnam, R. B. (2000). *The collapse and revival of American community*. New York: Simon and Schuster.
- Rahman, Z., Msadek, S., & Waly., H. (2002). Report in the observance of standards and codes: Accounting and auditing ROSC. *World Bank*, Cairo: Arab Republic of Egypt, 2–17.
- Rajan, R. G., & Zingales, L. (2003). The great reversals: The politics of financial development in the 20th century. *Journal of Financial Economics*, 69(1), 5–50.
- Rauch, J. (1995). Demo sclerosis: *The silent killer of American government*. New York: Random House.
- Reiter, S. (1997). The ethics of care and new paradigms for accounting practice. *Binghamton University Accounting Auditing & Accountability Journal*, 10(3), 299–324.
- Report on the observance of standards and codes (ROSC). 2001. Corporate Governance Country Assessment: Arab Republic of Egypt. *World Bank, IMF*, 2–41. Retrieved from: http://www.worldbank.org/ifa/Egyptrosc.pdf.
- Richard, C. (2006). Why an auditor can't be competent and independent: A French case study. *European Accounting Review*, 15(2), 153–179.
- Richardson, A. (2009). Regulatory networks for accounting and auditing standards: A social network analysis of Canadian and International Standard-Setting. *Accounting Organizations and Society*, 34(4), 571–588.
- Riyatno. (2007). Public accounting firm size effect on earnings response coefficients. *Journal of Financial and Business*, 5(2).

- Roberts, J. (2004). *The modern firm: Organizational design for performance and growth*. New York: Oxford University Press.
- Robertson, J.C., & Hawkins, K. (1999). Certified public accountants' self-regulation in socio-legal perspective. *Research in Accounting Regulation*, 13, 55–80.
- Rockness, H., & Rockness, J. (2005). Legislated ethics: from Enron to Sarbanes-Oxley, the impact on corporate America. *Journal of Business Ethics*, 57(1), 31–54.
- Romano, R. (2005). The Sarbanes-Oxley Act and the making of quack corporate governance. *Yale Law Journal*, 114(7), 1521–1550.
- Romanus, R. N., Maher, J., & Fleming, D. M. (2008). Auditor industry specialization and accounting restatements. *American Accounting Association*, 22(4), 1–5.
- Rona-Tas, A., & Hiss, S. (2010). The role of ratings in the subprime mortgage crisis: the art of corporate and the science of consumer credit rating. *Research in the So-ciology of Organizations*, 30a, 115–155.
- Ronen, J., & Yaari, V. L. (2008). *Earnings management emerging insights in theory, practice, and research*. NY: Springer Science and Business Media.
- Roychowdhury, S. (2006). Earnings management through real activities manipulation. *Journal of Accounting and Economics*, 42(3), 335–370.
- Sackman, S.A. (1991). *Cultural knowledge on organisations: Exploring the collective mind*. California: Sage Publications, Inc.
- Satava, D., Caldwell, C., & Richards, L. (2006). Ethics and the auditing culture: Rethinking the foundation of accounting and auditing. *Journal of Business Ethics*, 64(3), 271–284.
- Schein, E. H. (3rd. ed.). (2004). *Organizational culture and leadership*. San Francisco: Jossey-Bass a Wiley Imprint.

Schermerhorn, J.R. (2nd. Ed.). (2010). Exploring Management. John Wiley & Sons.

- Schipper, K. (1989). Commentary on earnings management. *Accounting Horizons*, 3(4), 91–102.
- Schuler, R. S. (1975). Role perceptions, satisfaction, and performance: A partial reconciliation. *Journal of Applied Psychology*, 60(6), 683–687.
- Scott, R. (2003). *Organizations: Rational, natural and open systems*. NJ: Pearson Upper Saddle River.
- Scott, W. R. (1995). Institutions and organizations. Thousand Oaks, Sage.
- Shafer, W. E., Morris, R. E., & Ketchand, A. A. (2001). Effects of personal values on auditors' ethical decisions. *Journal of Business Ethics*, 3(1), 243–260.
- Shapiro, A. (2011). Who pays the auditor calls the tune? Auditing regulation and clients incentives. *Seton Hall Law Reviewer Series*, 35(3), 1030–1056.
- Shawvere, T. J., & Barre, W. (2006). An exploratory study assessing the effectiveness of a professional responsibility course. *Global Perspectives on Accounting Education*, 3, 49–66.
- Sikka, P. (2009). Financial crisis and the silence of the auditors. *Accounting Organizations and Society*, 34(6), 868–873.
- Sillince, J. A. A. (2005). A contingency theory of rhetorical congruence. *Aston University Academy of Management Review*, 30(3), 608–621.
- Simon, D.T. (2011). Additional evidence on the large audit fee premium as an indication of audit quality. *Journal of Applied Business Research*, 13(4), 21–30.
- Tay, J.S., & Parker, R. H. (1990). Measuring harmonization and standardization. *Abacus*, 26(1), 73–85.
- Taylor, J. B. (2009). The financial crisis and the policy responses: An empirical analysis of What Went Wrong. *National Bureau of Economic Research*, 1–3. Retrieved from: http://www.nber.org/papers/w14631.pdf

- The Egyptian Ministry of Foreign Trade. (2003). *The Egyptian Standards on Auditing Implementation*, 25–26.
- Thompson, J. D. (1967). Organizations in action. New York: McGraw-Hill.
- Todaro, M. P. & Smith, S. C. (11th. ed.). (2011). *Economic Development*. Pearson Education.
- Trice, H., & Beyer, J. (1993). *The cultures of work organizations*. N.J: Upper Saddle River Prentice Hall.
- Tunstall, W.B. (1985 b). Breakup the bell system: A case study in cultural transformation. San Francisco: Jossey-Bass Publisher.
- Unerman, J., & O'Dwyer, B. (2004). Enron, WorldCom, Andersen et al: A challenge to modernity. *Critical Perspectives on Accounting*, 15(6–7), 971–993.
- Van de Ven. (1985). Alternative forms of fit in contingency theory. *Administrative Science Quarterly*, 30(4), 514–539.
- Van Lent, L. (1999). The economics of an audit firm: The benefits of partnership governance. *The British Accounting Review*, 31(2), 225–254.
- Violet, W. J. (1983). The development of international accounting standards: an anthropological perspective. *The International Journal of Accounting Education and Research*, 18(2), 1–12.
- Virovere, A. & Rihma, M. (1997). Ethics auditing and conflict analysis as management tools. School of Economics and Business Administration: TUTWPE, 67:77. Retrieved from: http://ideas.repec.org/p/ttu/wpaper/184.html
- Vona, L. W. (2008). Fraud risk assessment: Building a fraud audit program. John Wiley & Sons.
- Wahdan, M. A., Spronck, P., Ali, F. H., Vaassen, E., & Herik, H. J. (2005). Auditing in Egypt: A study of the legal framework and professional standards. *Maastricht School of Management*,1-10. Retrieved from: http://ilk.uvt.nl/~pspronck/pubs/Wahdan2005a.pdf

- Ward, G. (2004). The importance of professionalism in globally interdependent markets. International Federation of Accountants, 2–7. Retrieved from: http://www.actuaries.org.uk/system/files/documents/pdf/theimportanceofprofessionalismingloballyinterdependentmarkets.pdf
- Whittington, R. (1988). Environmental structure and theories of strategic choice. *Journal of Management Studies*, 25(6), 523–531.
- Windsor, C. & Warming-Rasmussen, B. (2009). The rise of regulatory capitalism and the decline of auditor independence: A critical and experimental examination of auditors' conflicts of interests. *Critical Perspectives on Accounting*, 20(2), 270–279.
- Zeghal, D. and Mhedhbi, K. (2006). An analysis affecting the adoption of international accounting standards by developing countries. *International Journal of Accounting*, 41(4), 373–386.
- Zhang, I. (2007). Economic consequences of the Sarbanes–Oxley Act of 2002. *Journal* of Accounting and Economics, 44, 74–88.
- Zucker, L. G. (1986). Production of trust: Institutional sources of economic structure. *Greenwich Research in organizational behavior*, 8, 62–94.

- Aggarwal, R., & Williamson, R. (2006, February). Did new regulations target the relevant corporate governance attributes? *McDonough School of Business, Georgetown University*. Washington DC, 2–9.
- Agoglia, C. P., Doupnik, T. S., & Tsakumis, G. T., (2011). Principles-based versus Rules-based Accounting Standards: The influence of Standard Precision and Audit Committee Strength on Financial Reporting Decisions. *The Accounting Review*, 86(3), 747–767.
- Agrawal, A. & Cooper, T. (2009, January). Corporate governance consequences of accounting scandals: evidence from top management, CFO and auditor turnover. 2nd Annual Conference on Empirical Legal Studies Paper, San Francisco, 2–7.
- Allred, B. & Swan, S. (2005). The mediating role of innovation on the influence of industry structure and national context on firm performance. *Journal of International Management*, 11, 232–242.
- Almer, R, & Brody, R. (2002). An empirical investigation of context-dependent communications between auditors and bankers. *Managerial Auditing Journal*. 17(8), 478–487.
- Ambrosini, V. & Bowman, C. (2003). Managerial consensus and corporate strategy: Why do executives agree or disagree about corporate strategy? *European Management Journal*, 21(2), 215–220.
- Andrews, R., Boyne, G. A., & Walker, R. M. (2006). Strategy content and organizational performance: an empirical analysis. *Public Administration Review*, 66(1), 54–60.
- Ashbaugh, H., LaFond, R. & Mayhew, B. W. (2003). Do Non-audit services compromise auditor independence? Further evidence. *The Accounting Review*, 78(3), 611–39.

- Ashbaugh-Skaife, H., Collins, D., Kinney Jr. W., & Lafond, R. (2009). The effect of SOX internal control deficiencies on firm risk and cost of equity. *Journal of Accounting Research*, 47(1), 1–17.
- Atuahene-Gima, K., & Li, H. (2004). Strategic decision comprehensiveness and new product development outcomes in new technology ventures. Academy of Management Journal, 47(4), 583–597.
- Barber, B. (1983). *The logic and limits of trust*. New Brunswick, NJ: Rutgers University Press.
- Barnard, I. C. (1938). *The functions of the executive*. Cambridge: Harvard University Press.
- Barth, J. R., Li, Tong, Phumiwasana, W. T., & Yago, G. (2009). *The rise and fall of the US mortgage and credit markets: A comprehensive analysis of the market meltdown*. New York: John Wiley and Sons.
- Barth, M.E. & Hutton, A. P. (2004, March). Analyst earnings forecast revisions and the pricing of accruals. *Review of Accounting Studies*, 9, 59–96.
- Barzel, Y. (1982). Measurement cost and the organization of the market. *Journal of Law and Economics*, 15, 27–48.
- Beneish, M. D., Hopkins, P. E., & Jansen, I. Ph. (2001). Do auditor resignations convey private information about continuing audit clients? *Kelley School of Business*, 3–25. Retrieved from: http://papers.ssrn.com/sol3/papers.cfm?abstract_id=268953
- Beretta, S. & Bozzolan, S. (2004). A framework for the analysis of firm risk communication. *The International Journal of Accounting*, 39, 265–288.
- Bicksler, J. L. (2008). The subprime mortgage debacle and its linkages to corporate. *International Journal of Disclosure and Governance*, 5(4), 295–300.

- Brouthers, K. D., Brouthers, L. E., & Werner, S. (2000). Influences on strategic decision making in the Dutch financial services industry. *Journal of Management*, 26(5), 863-879. doi: 10.1177/014920630002600506
- Browning, E. S., & Weil, J. (2002, January 29). Accounting woes roil stock markets as nervous investors stampede exits. *The Wall Street Journal*. Retrieved from: http://online.wsj.com/article/SB1012348529193369880.djm.htl
- Brunnermeier, M. K. (2009). Deciphering the liquidity and credit crunch 2007–2008. Journal of Economic Perspectives, 23(1), 79–90.
- Burns, N., & Kedia, S. (2006). The impact of performance-based compensation on misreporting. *Journal of Financial Economics*, 79, 35–67.
- Callen, J. L., & M. Morel. (2002). The Enron-Andersen debacle: Do equity markets react to auditor reputation? *Rotman School of Business, University of Toronto*, Working paper.
- Chaney, P. K., & Philipich, K. L. (2002). Shredded reputation: The cost of audit failure. *Journal of Accounting Research*, 40(4), 1221–1245.
- Chung, H. S., & Kallapur, S. (2003). Client importance, Non-audit services, and abnormal accruals. *The Accounting Review*, 78(4), 931–955.
- Cohen, D., Dey, A., & Lys, T. (2008, May). Real and accrual based earnings management in the Pre and post Sarbanes Oxley Periods. *The Accounting Review*, 83, 757–787.
- Coleman, C. & Bryan, C. L. (2002, August). Audit fees rise, and investors may pay price. *The Wall Street Journal*. Retrieved from: http://online.wsj.com/news/articles/SB1029097037499427475
- Crewson, P. E. (1997). Public service motivation: building empirical evidence of incidence and effect. *Journal of Public Administration Research and Theory*, 7(4), 499–518.

- Cullinan, C. (2004). Enron as a symptom of audit process breakdown: Can the Sarbanes- Oxley Act cure the disease? *Critical Perspectives on Accounting*, 15, 853–864.
- Cunningham, L. A. (2004). A new product for the state corporation law market: audit committee certifications. *Boston College Law School Public Law and Legal Theory*, 21–25.
- Cunningham, G. M. & Harris, J. E. (2006). Enron and Arthur Andersen: The case of the crooked E and the fallen A. *Global Perspectives on Accounting Education*, 3, 29–42.
- Daft, R. L., & Weick, K. E. (1984). Toward a model of organizations as interpretation systems. *Management Science Magazine Review*, 9, 284–295.
- Daske, H., & Gebhardt, G. (2006). International financial reporting standards and experts' perceptions of disclosure quality. *Abacus*, 42(3–4), 461–498.
- Daugherty, B.E., & Tervo, W.A. (2008). Auditor changes and audit satisfaction: Client perceptions in the Sarbanes-Oxley Era of legislative restrictions and involuntary auditor change. *Critical Perspectives on Accounting*, 19(7), 931–951.
- Davidson, R. H. (2011). Accounting fraud: booms, busts, and incentives to perform. University of Chicago Booth School of Business, 2–5.
- Davies, H., & Walters, P. (2004). Emergent patterns of strategy, environment and performance in a transition economy. *Strategic Management Journal*, 25, 348– 356.
- DeAngelo, L. (1981). Auditor size and auditor quality. *Journal of Accounting and Economics*, 3, 183–199.
- Drucker, P. E. (1954). The practice of management. New York: Harper & Row.
- European Commission. (2008, November 12). Commission staff working document accompanying the proposal for a regulation of the European Parliament and of

the Council on Credit rating Agencies – Impact Assessment. SEC/2008/2746 final, 5–7. Retrieved from: http://ec.europa.eu/internal_market/securi-ties/docs/agencies/impact_assesment_en.pdf

- Ewert, R. & Wagenhofer, A. (2004). Economic effects of tightening accounting standards to restrict earnings management. *The Accounting Review*, 80(4), 1101– 1124.
- Fama, E.F., & Jensen, M.C. (1983a). Separation of ownership and control. *Journal* of Law and Economics, 26, 301–325.
- FASB. (2009, April). FASB Issues Final Staff Positions to Improve Guidance and Disclosures on Fair Value Measurements and Impairments. *FASB News Release*, 2. Retrieved from: http://www.fasb.org/news/nr040909.shtml
- Ferguson, N., *The ascent of money: A financial history of the world*. London: Allen Lane. 2008.
- Frecka, T. J., & Nichols, W. D. (2004). Characteristics of master's in accounting degree programs. *Issues in Accounting Education*, 19(2), 165–189.
- Friedman, T. L. (2005). *The world is flat: A brief history of the twenty-first century*. NY: Douglas & McIntyre.
- Gaynor, L. (2002). Investors' perceptions of auditor independence: The effect of disclosure versus auditor restrictions. *Working paper, Georgetown University*, 2–5.
- Getzen, T.E. A. (1984). Brand name firm theory of medical group practice. *The Journal* of *Industrial Economics*, 33(2), 199–215.
- Graham, J., Harvey, C., & Rajgopal, S. (2005). The economic implications of corporate financial reporting. *Journal of Accounting and Economics*, 40(1), 3–45. Retrieved from: http://www.nber.org/papers/w10550

- Gul, F.A., Srinidhi, B., & Shich, T. (2002). The Asian financial crisis, accounting conservatism and audit fees: evidence from Hong Kong. Social Science Electronic Publishing, 2–8. Retrieved from: http://papers.ssrn.com/sol3/papers.cfm?abstract_id=315062
- Healy, P. M., & K. G. Palepu. (2001). A review of the empirical disclosure literature. *Journal of Accounting and Economics*, 31(3), 405–440.
- Heninger, W. G. (2001). The association between auditor litigation and abnormal accruals. *The Accounting Review*, 76(1), 111–126. doi: 10.2308/accr.2001.76.1.111
- Hermalin, B. (2005). Trends in corporate governance. *Journal of Finance*, 60(5), 2352–2356.
- Hildyard, N. (2008). A Crumbling wall of money financial bricolage, derivatives, and power. The Corner House, 1–23. www.thecornerhouse.org.uk/pdf/document/WallMoneyOct08.pdf>.
- Hogan, C. E. & Jeter, D. C. (1999). Industry specialization by auditors. *Auditing Journal of Practice and Theory*, 15(1), 1–9.
- Hosmer, L. T. (1994). Why be moral? A Different rationale for managers. *Business Ethics Quarterly*, 4(2), 193–202.
- Husted, B. W. (1994). Honor among thieves: A Transaction-Cost approach to corruption in Third-World countries. *Business Ethics Quarterly*, 4(1), 19–22.
- Ireland, J.C., & Lennox, C.S. (2000). The large audit firm fee premium: a case of selectivity bias. *Journal of Accounting, Auditing & Finance*, 17(1), 73–91.
- Jain, P., Kim, J., & Rezaee, Z. (2008). The Sarbanes-Oxley Act of 2002 and market liquidity. *Financial Review*, 43, 372–382.
- Jelinek, M. (1977). Technology, organizations, and contingency. *Academy of Management Review*, 2, 1–17.

- Jensen, M. C. (2002). Value maximization, stakeholder theory, and the corporate objective function. *Business Ethics Quarterly*, 12(2), 10–13.
- Jensen, M.C., Murphy, K., & Wruck, E. (2004). Remuneration: Where we've been, how we got to here, what are the problems, and how to fix them. *Harvard Business School NOM Research Paper*, No. 04-28, 44–77. Retrieved from: http://www.cgscenter.org/library/board/remuneration.pdf
- Johnston, J. & Madura, J. (2009). The pricing of IPOs post-Sarbanes-Oxley. *Financial Review*, 44(2), 291–310.
- Keats, B. (1983). On the measurement of strategic performance. The Strategic Management Society Conference, Paris. Retrieved from: http://strategicmanagement.net/pub/index.php
- Kim, J., & Yi, H. (2006). Ownership structure, business group affiliation, listing status and EM: evidence from Korea. *Contemporary Accounting Research*, 23(2), 427–464.
- Klein, B. (1977). Dynamic economics. Harvard University Press Management Review.
- Krishnan, G., & Parsons, L. (2008). Getting to the bottom line: An exploration of gender and earnings quality. *Journal of Business Ethics*, 78(1), 65–76.
- Laird, W. M., & Dennis, L. (1979). An alternative to macro-micro contingency theories: An integrative model. *Academy of Management Review*, 4(3), 333–345.
- Landa, J. T. (1994). *Trust ethnicity, and identity: Beyond the new institutional economics of ethnic trading networks, contract law, and gift-exchange.* University of Michigan Press.
- Landsman, W.R., Nelson, K.K. and Rountree, B.R. (2009). Auditor switches in the preand post-Enron eras: Risk or realignment? *The Accounting Review*, 84(2), 531–558.

- Langohr, H., & Langohr, P. (2008). *The rating agencies and their credit ratings: What they are, how they work, and why they are relevant*. UK: John Wiley and Sons.
- Leiblein, M. J., & Miller, D. J. (2003). An empirical examination of transaction and firm-level influences on the vertical boundaries of the firm. *Strategic Management Journal*, 24, 839–859.
- Magnan, M. (2009). Fair value accounting and the financial crisis: Messenger or contributor? *Accounting Perspectives*, 8(3), 189–213.
- Maijoor, S., & Meuwissen, R. (1993). Mobility of auditors and the nature of audit services: Exploratory evidence from the Dutch audit market. *Maastricht University Accounting and Auditing Research and Education Center*, 2–14.
- McCracken, S., Salterio, S. E., & Gibbins, M. (2008). Auditor-client management relationship and roles in negotiating financial reporting. *Accounting, Organizations and Society*, 33(4/5), 362–83.
- McDaniel, R. W. (2008). Written statement before the United States Securities and Exchange Commission SEC. Retrieved from: http://www.sec.gov/news/extra/credrate/moodys.htm
- Middlemiss, N. (2003). Authentic, not cosmetic: CSR as brand enhancement. *Journal of Brand Management*, 10(4), 355–360.
- Moncarz, E. S., Moncarz, R., Cabello A., & Moncarz B. (2006). The rise and collapse of Enron: Financial innovation, errors and lessons. *enero-abril*, 218, 3–17. Retrieved from: http://www.redalyc.org/articulo.oa?id=39521802&idp=1&cid=18683
- Montgomery, D., Beasley, M., Menelaides, S., & Palmrose, Z. (2002). Auditors' new procedures for detecting fraud. *Journal of Accountancy*, May: 63–66. Re-trieved from: http://www.journalofaccountancy.com/Issues/2002/May/AuditorsNewProceduresForDetectingFraud.htm

- Owhoso, V.E., Messier, W. F., & Lynch, J.G. (2002). Error detection by industry-specialized teams during sequential audit review. *Journal of Accounting Research*, 40(3), 883–900.
- Povel, P., Singh, R., & Winton, A. (2007). Booms, busts, and fraud. *Review of Financial Studies*, 20(4), 1219–1254.
- Powers, Jr. W.C., Troubh, R.S., & Winokur, Jr. H. S. (2002). Report on investigation by the special investigative committee of the board of directors of Enron Corp. Austin: Wilmer, Cutler & Picketing. Retrieved from: http://i.cnn.net/cnn/2002/LAW/02/02/enron.report/powers.report.pdf
- Riotto, J. (2008). Understanding the Sarbanes-Oxley Act—a value added approach for public interest. *Critical Perspectives on Accounting*, 19(7), 952–962.
- Rousseau, S. (2006). Enhancing the accountability of credit rating agencies: The case for a disclosure based approach. *McGill Law Journal*, 51, 617–628. Retrieved from: http://www.lawjournal.mcgill.ca/userfiles/other/5366167-Rousseau1.pdf
- Roy A. C. (1999). Critics use of satire against the auditing profession: A short historical note. *Cardiff Business School Accounting Auditing & Accountability Journal*, 12(1), 129–133.
- Rudnicki, M. (2000, November 10). *Environmental and social responsibility as the value in modern politics. Paper delivered during meeting*. The OIKOS International. Warsaw.
- SEC, Securities and Exchange Commission. (2008), Summary report of issues identified in the commission staff's examinations of select credit rating agencies by the staff of the Securities and Exchange Commission. 23–32 Retrieved from: http://www.sec.gov/news/studies/2008/craexamination070808.pdf.
- Shafer, W. E., Ketchand, A. A., & Morris R. E. (2004). Auditors' willingness to advocate client-preferred accounting principles. *Journal of Business Ethics*, 52(3), 213–227.

- Shapiro, Amy. (2005). Who pays the auditor calls the tune? Auditing regulation and clients' incentives. Cornell Law School Legal Studies Research Paper Series, 30, 1066–1094.
- Solomon, R. C. (1992). Corporate roles, personal virtues: An Aristotelian approach to business ethics. *Business Ethics Quarterly*, 2(3), 317–340.
- Soros, G. (2008, January 22). The worst market crisis in 60 years. London: *Financial Times*. Retrieved from: http://www.ft.com/intl/cms/s/0/24f73610-c91e-11dc-9807-000077b07658.html#axzz2hiqRf2PM
- Stiglitz, J. (2003). The roaring nineties: Seeds of destruction. London: Penguin.
- Story, D., Price, T. J. (2006). Corporate social responsibility and risk management. *The Journal of Corporate Citizenship*, 22, 44–49.
- Taleb, N. N. (2nd ed.). (2007). *The black swan: The impact of the highly improbable*. USA: Random House.
- Tan, J., & Tan, D. (2005). Environment-strategy co-evolution and co-alignment: a staged model of Chinese SOEs under transition. *Strategic Management Journal*, 26(2), 143–152.
- Taylor, F. W. (1947). Scientific management. New York: Harper & Brothers.
- Thibodeau, J. C., & Freier, D. (2nd ed.). (2009). Auditing after Sarbanes-Oxley: Illustrative Cases. New York: McGraw-Hill Irwin.
- Tillema, S. (2005). Towards an integrated contingency framework for MAS sophistication: Case studies on the scope of accounting instruments in Dutch power and gas companies. *Management Accounting Research*, 16(1), 103–111.
- Tushman, M. L., & Nadler, D. A. (1978). Information processing as an integrating concept in organization structure. Academy of Management Review, 3(3), 613– 624.

- Van, T. B., & Vanstraelen, A. (2005). Earnings management under German GAAP versus IFRS. *European Accounting Review*, 14(1), 155–180.
- Volcker, P. A. (2002, November 19). Accounting, accountants, and accountability in an integrated world economy. The World Congress of Accountants. Hong Kong, 1–5 Retrieved from: http://www.iasplus.com/en/binary/resource/volcker0211.pdf.
- Yu, Hung-Chao. (2004). Legal regimes, damage apportionment rules, and auditor independence theory and experimental evidence. *College of Commerce, National Chengchi University*, 1–16. doi: 10.2139/ssrn.463940.
- Zajac, E. J., Kraatz, M. S., & Bresser, R. F. (2000). Modeling the dynamics of strategic fit: a normative approach to strategic change. *Strategic Management Journal*, 21, 431–538.
- Zuniga-Vicente, J., & Vicente-Lorente, J. (2006). Strategic moves and organizational survival in turbulent environments: The Case of Spanish Banks. *Journal of Management Studies*, 43, 485–519.

List of Publications

List of Publications

Published Researches Details

Research Title: Canadian Monetary Policy. Magazine Name: Management Academic Researches Magazine Magazine Type: Academic Periodical Magazine Researches Types: Academically Accredited Researches Publisher: Sadat Academy for Management Science Periodic Numbers: Volume 23, Number 3 Research Title: University Role in Research and Economic Setting: Canadian Universities Innovations. Magazine Name: Management Academic Researches Magazine Magazine Type: Academic Periodical Magazine Researches Types: Academically Accredited Researches Publisher: Sadat Academy for Management Science Periodic Numbers: Volume 24, Number 4

Current Researches in Publishing Process Research Title: Audit Practicing and Financial Market Failures Magazine Name: ESAA Professional Researches Magazine

Magazine Name: ESAA Professional Researches Magazine Magazine Type: National Profession Periodical Magazine Researches Types: Paper Researches Publisher: Egyptian Society of Accountants and Auditors (ESAA) Research Title: **"Audit Professions and Enron Failure''** Magazine Name: ESAA Professional Researches Magazine Magazine Type: National Profession Periodical Magazine Researches Types: Paper Researches Publisher: Egyptian Society of Accountants and Auditors (ESAA)

Subscription in Published Book

Book Name: Interpretations of International Financial Reporting Standards. Authors: Nasim Awad, Salah Al- Alami, and Khaled Omar Publisher: Department of Publishing, State of Kuwait. 2007

Seminars

Seminar Title: GAAP & GAAS Review & Update Seminar Date: 2/10/2014 Seminar Administrators: CGA Ontario Professional Developments